

Temi di discussione

(Working Papers)

Demographics and the natural real interest rate: historical and projected paths for the euro area

by Andrea Papetti

I306



Temi di discussione

(Working Papers)

Demographics and the natural real interest rate: historical and projected paths for the euro area

by Andrea Papetti

Number 1306 - November 2020

The papers published in the Temi di discussione *series describe preliminary results and are made available to the public to encourage discussion and elicit comments.*

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Monica Andini, Audinga Baltrunaite, Marco Bottone, Davide Delle Monache, Sara Formai, Francesco Franceschi, Salvatore Lo Bello, Juho Taneli Makinen, Luca Metelli, Mario Pietrunti, Marco Savegnago. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

DEMOGRAPHICS AND THE NATURAL REAL INTEREST RATE: HISTORICAL AND PROJECTED PATHS FOR THE EURO AREA

by Andrea Papetti*

Abstract

This paper employs a large-scale overlapping generation (OLG) model quantifying that demographics account for a decrease in the natural real interest rate of about 1.4 percentage points in the euro area compared with the average for the 1980s to 2030 (roughly at its trough), under the baseline calibration. Two channels prevail in providing the downward impact: the increasing scarcity of effective labor input and the growing willingness of individuals to save due to longer life expectancy. Mitigating factors are: greater substitutability between labor and capital, higher intertemporal elasticity of substitution in consumption, higher productivity and participation by older individuals and, to a lesser extent, a higher retirement age. Absent pay-as-you-go pension systems, the natural rate would stand at a lower level of about 0.5 percentage points by 2030. The simulated paths of the natural real interest rate and output growth are consistent with econometric estimates.

JEL codes: E17, E21, E43, E52, J11

Keywords: population aging, natural interest rate, secular stagnation, overlapping generations **DOI:** 10.32057/0.TD.2020.1300

1. I	Introduction					
2. 0	OLG model10					
3. I	Understanding 1the channels: two-period version of the model					
4. (Quantitative analysis					
4	4.1 Experiment and baseline calibration					
4	4.2. Baseline results					
4	4.3. Isolating the drivers					
4	4.4. Sensitivity analysis					
	4.4.1	Intertemporal elasticity of substitution	28			
	4.4.2	Capital-labor substitutability	30			
	4.4.3	Age-dependent labor efficiency	32			
	4.4.4	Type of pension system				
	4.4.5	Retirement age	34			
5. (Comparing	the mode with the data: the role of TFP growth				
	5.1 Experi	ment and calibration with non-zero TFP growth				
	5.2 Results	: model vs data	40			
	5.3 Isolatin	ng the drivers	43			
6. (6. Concluding remarks					
References						
App	Appendix A: Additional figures					
Appendix B: OLG model with zero technology growth						
Appendix C: OLG model with non-zero TFP growth						
Appendix D: Two-period version of the model						
Appendix E: Solow residual in the euro area						
Appendix F: Measures of real interest rate in the euro area						
Appendix G: Real interest rate: further comparison of the model results with data						
Appendix H: Robustness: alternative to perfect annuity market						

Contents

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research, *Research Fellowship*.

We have this unusual degree of knowledge concerning the future because of the long but definite time-lag in the effects of vital statistics. Nevertheless the idea of the future being different from the present is so repugnant to our conventional modes of thought and behaviour that we, most of us, offer a great resistance to acting on it in practice. (Keynes (1937), Some Economic Consequences of a Declining Population)

A temporary period of policy rates being close to zero or even negative in real terms is not unprecedented by any means. Over the past decades, however, we have seen longterm yields trending down in real terms as well, independent of the cyclical stance of monetary policy. (Draghi (2016), Addressing the Causes of Low Interest Rates)

1 Introduction¹

Advanced economies are undergoing a demographic transition, the ageing process by which "populations move from initially high fertility and mortality with young age distributions to low fertilty and mortality with old age distributions" (Lee, 2016). As the number of people entering the world is shrinking and mortality rates are decreasing (i.e. the survival probabilities are increasing), the relative number of the elderly is dramatically increasing. While before the 1980s the ratio of the elderly (aged 65 and over) to working age (aged 15-64) has been less than 2 to 10, the United Nations (UN, 2017) project this proportion to rise above 5 to 10 by year 2050 in Europe.² Based on definite time-lags, demographic projections offer a relatively reliable knowledge of the future. It is therefore appealing to use demographic data as exogenous variation to explain macroeconomic dynamics.

Questioning the influence of demographic change on the real interest rate is certainly not new in economic research. What is new in recent years is that the topic is on the agenda of central bankers. The fact that real interest rates have been on a downward trend since the late 1980s across many countries leads to ask whether the natural interest rate has decreased as well and whether it will remain low in the years ahead, potentially hampering the effectiveness of monetary policy.

The definition of natural or neutral interest rate dates back to Wicksell (1898)³ and in modern macroeconomics can be thought as the rate of interest that brings output in line with its potential or natural level in the absence of transitory shocks or nominal adjustment frictions (cf. Woodford (2003), Brand et al. (2018)).⁴ It will be identified as the real rate of return on capital (net of

¹I am particularly grateful to Claus Brand for his continuous support, to Carlos Montes-Galdón for routing suggestions in the early stages of the project, and to Lars Ljungqvist for invaluable guidance and advice. For helpful comments and discussions, I thank Andrea Ferrero, Jordi Galí, Callum Jones, Eliza Lis, Alexander Ludwig, Christiane Nickel, Chiara Osbat, Oreste Tristani, Leopold von Thadden, Gianluca Violante, seminar participants at the European Central Bank, Stockholm School of Economics, Swedish Ministry of Finance, Bank of Italy and team members of the Working Group on Econometric Modelling (WGEM) of the European System of Central Banks (ESCB). An earlier version of this paper circulated under the same title via https://www.ecb.europa.eu//pub/pdf/scpwps/ecb. wp2258~32d1cdba97.en.pdf. All errors are my own. This paper should not be reported as representing the views of the Bank of Italy or the European Central Bank (ECB) or the ESCB. The views expressed are my own and do not necessarily reflect those of the Bank of Italy or the ECB or the ESCB. *Postal address*: Bank of Italy – Directorate General for Economics, Statistics and Research – Via Nazionale 91, 00184 Rome (Italy). *E-mail*: papetti.andrea@gmail.com

²See Figure A.1 in the Appendix. The aging process is very similar across different European geographic areas, no matter if one considers the euro area composed by 19 countries, the big 5, the core 12, or the wide European Union composed by 28 countries.

³ "There is a certain rate of interest on loans which is neutral in respect to commodity prices, and tends neither to raise nor to lower them. This is necessarily the same as the rate of interest which would be determined by supply and demand if no use were made of money and all lending were effected in the form of real capital goods. It comes to much the same thing to describe it as the current value of the natural rate of interest on capital". (Wicksell, 1898)

⁴"That is, the interest rate must at all times equal the Wicksellian natural rate of interest, which may be defined as the equilibrium real rate of return in the case of fully flexible prices. Under this definition, one observes a direct correspondence with the previously introduced concept of the natural rate of output. Indeed, the natural rate of interest is just the real rate of interest required to keep aggregate demand equal at all times to the natural rate of output." (Woodford, 2003).

depreciation) that allows the saving supply (by households) to meet the capital demand (by firms) in absence of any allocational friction or arbitrage. The *real* macro model employed in the paper will abstract from frictions and shocks that could capture business cycle variations. The natural rate of return on capital will be shortly called real interest rate.

Researchers have been recently looking for potential "slow-moving secular forces" as explanatory factors behind the downward trend in real interest rates (cf. Eggertsson et al. (2019)). Demographic change is one of such forces and how it can affect the natural real interest rate, an *unobservable* variable, is the exclusive focus of this paper.

In a standard Solow (1956)'s model with homogeneous population and a constant saving rate, as population growth decreases capital per worker rises dampening the marginal product of capital, so that in equilibrium the more abundant factor, capital, receives a lower remuneration than the other factor, labor: the real interest rate falls while the real wage rises. Models embedded in the neoclassical framework, like general equilibrium overlapping generations (OLG) models, can rarely escape the prediction that ageing leads to a lower real interest rate.

Since the seminal contribution by Auerbach and Kotlikoff (1987), OLG models are considered the most reliable tool to evaluate the macroeconomic effect of demographic change as they allow to use the full empirical age distribution, in a context of a flexible life-cycle behavior. Recent contributions employing fully-fledged OLG models all predict a downward trend of the real interest rate due to population aging, no matter if the model encompasses the whole world economy with different countries/areas (cf. e.g. Domeij and Flodén (2006), Krueger and Ludwig (2007), Attanasio et al. (2007)) or a single country/area modeled as closed-economy (cf. e.g. Gagnon et al. (2016) for US, Bielecki et al. (2018, 2020) for Europe, Sudo and Takizuka (2019) for Japan).

This paper employs a large-scale OLG model, similar to those existing in the literature, to provide a quantification of the impact of the demographic transition on the aggregates for the euro area, with particular focus on the channels through which the real interest rate is affected. There are four channels (cf. Krueger and Ludwig (2007), and Carvalho et al. (2016)).

(1) Downward impact from increasing scarcity of labor supply. A decrease of the growth rate of the effective labor supply relative to the total population in the economy is akin to a slowdown in total factor productivity for output per capita growth, which leads firms to demand less capital, reducing the marginal product of capital and so the real interest rate, everything else being equal. This effect is stronger the more the age-distribution shifts towards older cohorts that are parametrized to be less productive and participative in the labor market, and the lower the degree of substitutability in production between capital and labor.

(2) *Downward impact from higher life expectancy*. With the goal of smoothing consumption into the future, households anticipate their higher survival probabilities with a willingness of consuming less and saving more, i.e. becoming more patient, thus decreasing the real interest rate, all else being equal. This effect depends on the way households are insured against mortality risk as well as on the pension scheme in place, and is stronger the lower the intertemporal elasticity of

substitution in consumption.

(3) Upward impact from a rising proportion of dissavers. According to the life-cycle theory embedded in the model, households build up their wealth during the most productive working ages and dissave in later ages. The shift of the population distribution towards relatively more dissavers along the transition dynamics can decrease the aggregate saving rate thus increasing the real interest rate by making capital scarcer, all else being equal.

(4) Upward impact from pension system crowding-out effect. A pay-as-you-go (PAYGO) pension system with defined-benefit, as often run in European countries in spite of its different hybrid reformulations, requires an increase of labor taxes for the government budget to be balanced as the relative pool of retirees increases. This tends to crowd out productive capital in the economy and, as capital becomes less abundant, to increase the real interest rate, all else being equal.

The model is calibrated at the annual frequency for the euro area using demographic data and projections by UN (2017) as exogenous variations to study a perfect-foresight transition where the demographic change is perfectly anticipated by the agents in the economy. The quantitative exercise shows that the real interest rate path guided by demographic change exhibits a slight rise throughout the 1970s and 1980s, and then a prolonged and marked fall at least until 2030 (when it roughly reaches its trough according to the demographic projections). The shape is consistent with low-frequency econometric estimates (see Holston et al. (2017), Fiorentini et al. (2018)).

In the baseline analysis the annual real interest rate declines about 1.4 percentage points going to 2030 compared to the average in the 1980s. Under the various sensitivity exercises the real interest rate is never found to reach a higher level compared to its initial steady state value in 1950. These exercises show that the dampening effect of aging could be mitigated not only by higher substitutability between labor and capital and higher elasticity of intertemporal substitution in consumption, but also by reforms aiming particularly at increasing the relative productivity of older cohorts and the participation rate. An increase of the retirement age with no other supporting reform leads to a limited increase in output per capita growth and to essentially no change in the real interest rate.

A decomposition of the main drivers in the model shows that the presence of the PAYGO pension system with defined-benefits is associated with an upward deviation of the real interest rate from the initial steady state of about 0.5 percentage point by 2030. By the same year the increase of the survival probabilities is associated with a negative deviation of more than 0.9 percentage points. Variations in the number of newborns in the model would account for the residual deviation of the real interest rate, about 0.5 percentage point, going to 2030 compared to the average in the 1980s.

Through the lens of this model, it does not seem to be really the case that demographics can meaningfully "reverse three multi-decade global trends" on real interest rates via the pension system channel as argued by Goodhart and Pradhan (2017). At least not until 2030. But then the system is predicted to stay at a permanent lower level of about 1 percentage point compared to

the one prevailed in the 1980s. Coherently with what recently found by Rachel and Summers (2019a,b), this paper supports the view that in absence of offsetting policies such as PAYGO pension benefits the natural real interest rate would have been at lower level and that, going forward, bigger policy shifts are needed if one wants to overturn the downward impact of aging.

Interestingly, the model predicts that before 2007 one could easily interpret the impact of demographic change on output growth as a type of scaling factor, producing if anything mild variations around a constant, close to zero for the case of output per capita. However, after 2007 – which, coincidentally, is when the "Global Financial Crisis" started – the model predicts a dramatic decrease of output growth due to demographic change.⁵ For the real interest rate too, the most dramatic phase is after 2007 with a variation of almost -1 percentage point between 2007 and 2030. Hence, what noticed by Gagnon et al. (2018) for the United States, employing a similar OLG model, seems also true for Europe: "the largest effects of demographics on interest rates and GDP growth coincide to some degree with the Global Financial Crisis". So that "downward pressures on interest rates and GDP growth due to demographics could be easily misinterpreted as persistent but ultimately temporary influences of the global financial crisis" (Gagnon et al. (2016), p. 3.). Instead, what OLG models seem to predict is a "new normal".

The quantitative results are not significantly affected when the model is augmented with exogenous time-varying total factor productivity (TFP) growth (inferred from a measure of the Solow residual). The associated re-calibration allows to be more consistent with the *level* of the econometric estimates of natural outcomes (real interest rate and output) as provided e.g. by Holston et al. (2017) while the *variation* that can attributed to aging remains of the same order of magnitude. Specifically, in the baseline with constant TFP growth in each period the real interest rate decreases by about 1.1 percentage points between 1990 and 2030, going from a value of about 2.24% in 1990 to 1.13% in 2030. Over the same period, the decrease is about 1.65 percentage points (2.22% to 0.57%) with the Hodrick-Prescott (HP) filtered TFP growth rate. Thus, the results from the OLG model allow to qualify the determinants of agnostic econometric estimates and provide a forecast relying on demographic projections. Compared to the average in the 1980s, the real interest rate decreases about 1.8 percentage points going to 2030. A decomposition shows that the progressive decline of TFP growth rates accounts for about 0.55 of those percentage points so that demographics account about two-thirds of the model's generated decline of the real interest rate.

This paper builds on a vast literature on large-scale OLG models, rapidly growing since the onset of the debate on "secular stagnation" (see Eggertsson et al. (2019)) where the closest match with the framework employed here can be probably found in Domeij and Flodén (2006), and Krueger and Ludwig (2007). The former focuses on capital flows in a multi-country setting, the latter is a seminal paper on the impact of aging on rates of return to capital providing quantitative estimates

⁵While in 2007 demographics was contributing to increase total output by 0.4% per annum, by 2015 the figure is -0.2%. Going to 2026 the model predicts a further drag on total output with an annual growth rate of about -0.6%. A similar persistent decrease is predicted for output per capita annual growth rate: it goes from -0.2% in 2007 to -0.4% in 2015, to -0.7% by 2026.

only from 2000. The purpose here is not to further refine their theoretical structure. Rather, to offer a quantification (for the euro area) of the total effect of aging as well as of the different channels in isolation in a model where the full age-structure of the population is allowed. Carvalho et al. (2016) study the channels above using the analytically-tractable set-up à *la* Gertler (1999) (applied to a representative OECD economy). They recognize that "tractability comes at the cost of not endowing the model with any flexibility to match the empirical age distribution", and "leave for future research an extensive comparison of the results obtained in this framework with those from a large-scale OLG model, disciplined by a richer set of moments from demographics data" (Carvalho et al. (2016), p. 212). Indeed, this paper tires to accomplish that need of future research.

Quantitative estimates for the euro area are provided by Kara and von Thadden (2016) using a Gertler (1999)'s type of model and by Bielecki et al. (2018, 2020) using a richer OLG model. Their results on the impact of demographic change on the real interest rate are of the same order of magnitude of those provided in this paper, even though there the focus is more on the interaction of demographic change with monetary policy in a New-Keynesian framework and with global factors.

The rest of the paper is organized as follows. Section 2 describes the OLG model environment. Section 3 studies the simplified two-period version of the model to understand and isolate analytically the channels. Section 4 provides the full set of quantitative analyses including calibration, description of the experiment to study the transition dynamics, and a sensitivity analysis of the results to different parameter values. Section 5 introduces non-zero total-factor-productivity (TFP) growth in the model and makes a comparison with econometric estimates. Section 6 concludes.

2 OLG model

Consider a model in the spirit of Domeij and Flodén (2006), Krueger and Ludwig (2007) for a closed-economy populated by overlapping generations (OLG) of households that solve a standard life-cycle consumption problem. The demographic development is exogenous.

Households. Each household consists of a single individual. Households within each cohort j are identical and their exogenous mass $N_{t,j}$ for time-period t evolves recursively according to:

$$N_{t,j} = N_{t-1,j-1} s_{t,j} \tag{2.1}$$

where $s_{t,j}$ is the conditional survival probability.⁶

A representative j-aged household entering the world at time t maximizes the utility function choosing consumption and the amount of assets to hold the following period for each life period,

⁶Given that an individual is aged j - 1 at time t - 1, $s_{t,j}$ is the probability to be alive at age j at time t. Following Domeij and Flodén (2006), data are taken for $N_{t,j}$ for all available t, j to get the implied survival probabilities $s_{t,j}$ which therefore can exceed 1 due to migration flows. The underlying assumption is that immigrants enter the economy without assets and are adopted by domestic households: assets are carried over between periods by a domestic cohort and then split among its survivors and the asset-less immigrants in the same age class.

 $c_{t+j,j}, a_{t+j+1,j+1}$. Assuming a CRRA utility function, the maximization problem is:⁷

$$\max_{c_{t+j,j}, a_{t+j+1,j+1}} \sum_{j=0}^{J} \beta^j \pi_{t+j,j} \frac{(c_{t+j,j})^{1-\sigma}}{1-\sigma}$$
(2.2)

subject to

$$a_{t+j+1,j+1} = \frac{a_{t+j,j}(1+r_{t+j})}{s_{t+j,j}} - c_{t+j,j} + y_{t+j,j}$$
(2.3)

$$y_{t+j,j} = (1 - \tau_{t+j}) w_{t+j} h_{t+j,j} I(j < j_r) + d_{t+j,j} I(j \ge j_r)$$
(2.4)

$$a_{t+J+1,J+1} = 0 (2.5)$$

$$a_{t,0} = 0$$
 (2.6)

where $\pi_{t+j,j} = \prod_{k=0}^{j} s_{t+k,k}$ is the unconditional survival probability with $s_{t,0} = 1$; β is the discount factor; σ is the coefficient of risk-aversion (here the inverse of the intertemporal elasticity of substitution in consumption); r_{t+j} is the real interest rate; w_{t+j} is the real wage; τ_{t+j} is the tax rate on labor income; $I(\cdot)$ is an indicator function; j_r denotes the retirement age, exogenously imposed; $h_{t+j,j} = h_j$ for all t is an exogenously given amount of hours to work depending on age, constant over time (cf. Figure A.3); $d_{t+j,j}$ is the pension transfer from the government. Each household is born with zero wealth and is not allowed to die with either positive or negative wealth. It is assumed that there exists a "perfect annuity market".⁸

Firms. Households have direct ownership of physical capital and rents it to the firms. Output (Y_t) is produced under perfect competition with constant elasticity of substitution (CES) technol-

$$a_{t+1,j+1} = a_{t,j}(1+r_t) + \frac{a_{t,j}(1+r_t)(1-s_{t,j})N_{t-1,j-1}}{N_{t-1,j-1}s_{t,j}} - c_{t,j} + y_{t,j}$$
$$= \frac{a_{t,j}(1+r_t)}{s_{t,j}} - c_{t,j} + y_{t,j}$$

⁷Notice that it is assumed that labor supply is exogenous. Of course, this a limitation of the analysis. However, similar OLG settings generally find small effects of introducing endogenous labour supply. For example, in a robustness analysis (section 6) of their seminal contribution, Attanasio et al. (2007) have: "The main finding is [that] endogenizing labor supply has little effect on the equilibrium". Therefore, it was opted to keep the analysis easier, not allowing for endogenous labor supply.

⁸The assumption of "perfect annuity market" means that the agents within each age group j agree to share the assets of the dying members of their age group among the surviving members, equally. Using the notation just introduced, consider those that at time t are aged j. The total amount of assets of the dying members is: $a_{t,j}(1 - s_{t,j})N_{t-1,j-1}$, while the number of surviving members is: $N_{t,j} = N_{t-1,j-1}s_{t,j}$. Hence, in the budget constraint the asset holding in period t + 1 will depend on what as been accumulated plus this sort of 'equal gift' from the dying members given the real interest rate (r_t) at which these assets can be invested (minus consumption plus income):

which is the budget constraint written in the main text. Appendix H provides a different model structure and results where the assumption of "perfect annuity market" is relaxed. Instead, it is assumed that accidental bequests resulting from premature death are taxed by the government at a confiscatory rate and used for otherwise neutral government consumption.

ogy:⁹

$$Y_t = \left[\psi(K_t)^{\frac{\rho-1}{\rho}} + (1-\psi)(A_tL_t)^{\frac{\rho-1}{\rho}}\right]^{\frac{\rho}{\rho-1}}$$
(2.7)

where $0 < \psi < 1$ is the capital share parameter, $0 < \rho < 1$ is the elasticity of substitution between capital and labor, A_t is the labor-augmenting technology assumed to grow exogenously: $A_t = g_t^A A_{t-1}$. The factor markets are also perfectly competitive. Therefore, one can consider a representative firm hiring (efficiency units of) labor L_t at a given hourly real wage w_t and renting capital K_t at the rental rate $r + \delta$, subject to the yearly depreciation rate δ . The representative firm maximizes per-period profits:

$$\max_{L_t, K_t} \{ Y_t - w_t L_t - (r_t + \delta) K_t \}$$
(2.8)

Government. Given a certain level of generosity of the pay-as-you-go (PAYGO) pension system, i.e. the replacement rate \bar{d} defined as the pension benefit d_t received by each household per unit of the average labor income $w_t(1 - \tau_t)\bar{h}$, the government sets a tax rate τ_t such that its budget is balanced in each period:¹⁰

$$d_t = \bar{d}w_t(1-\tau_t)\bar{h} \tag{2.9}$$

$$\tau_t w_t L_t = d_t \sum_{j=j_r}^{s} N_{t,j}$$
 (2.10)

Clearing. The factor markets and the goods market clear:

$$L_t = \sum_{j=0}^{J} h_j N_{t,j}$$
 (2.11)

$$K_{t+1} = \sum_{j=0}^{J} a_{t+1,j+1} N_{t,j}$$
(2.12)

$$C_t + I_t = Y_t \tag{2.13}$$

where $C_t = \sum_{j=0}^{J} c_{t,j}$ is aggregate consumption, while investment I_t satisfies the low of motion of capital: $K_{t+1} = (1 - \delta)K_t + I_t$.

Equilibrium. Given the exogenous demographic development (fully characterized by the incoming cohort size $N_{t,0}$ and the conditional survival probabilities $s_{t,j}$ according to (2.1)) and the labor-augmenting technology (A_t) in all periods $t = 0, 1, ..., \infty$ for all cohorts j = 0, 1, ..., J, the equilibrium for this (closed, perfectly competitive) economy is a sequence of prices $\{w_t, r_t\}_{t=0}^{\infty}$, policies $\{\tau_t\}_{t=0}^{\infty}$, transfers $\{d_t\}_{t=0}^{\infty}$ and quantities $\{\{c_{t,j}, a_{t,j}\}_{j=0}^{J}, K_t, L_t, Y_t, C_t, I_t\}_{t=0}^{\infty}$ such that:

 $^{^9 {\}rm For} \ \rho \rightarrow 1$ the function is of the Cobb-Douglas form.

¹⁰Have: $\bar{h} = \sum_{j=0}^{j_r-1} h_j / j_r$.

- 1. Households solve the optimization problem (2.2) subject to constraints (2.3)–(2.6);
- 2. firms maximize profits solving problem (2.8) given the production function (2.7);
- 3. the fiscal authority sets a tax rate (2.10) such that its budget is balanced in each period given the individual pension transfer (2.9);
- 4. factor markets (2.11), (2.12) and the goods market (2.13) clear satisfying the low of motion of capital: $K_{t+1} = (1 \delta)K_t + I_t$, with aggregate consumption $C_t = \sum_{j=0}^J N_{t,j}c_{t,j}$.

3 Understanding the channels: two-period version of the model

Consider the special case of the model above in which the representative households lives only for two periods, at j = 0 as a worker and at j = 1 as a retiree. Under this simple setting it is possible to characterize the equilibria in the model analytically, revealing the main channels through which demographic change affects the real interest rate, thus providing some intuition for the quantitative results from the full model simulation provided in the next sections.¹¹

Environment. Assume perfect foresight. Given the number of people in the working age in period t, $N_{t,0}$, and the probability to survive the next period (becoming a retiree), $s_{t+1,1}$, the number of old the subsequent period is given by: $N_{t+1,1} = s_{t+1,1}N_{t,0}$. A household born in period t has preferences over consumption $c_{t,0}$, $c_{t+1,1}$ representable by the utility function (consider the special case of log-preferences for simplicity, i.e. $\sigma = 1$):

$$\log(c_{t,0}) + \beta s_{t+1,1} \log(c_{t+1,1})$$

Households supply labor inelastically in the first period receiving a wage w_t . They retire in the second period receiving social security benefits d_{t+1} financed by payroll taxes on labor income. The holding of capital $(a_{t+1,1})$ gives a return r_{t+1} while the unintentional bequest is equally split among the survivors. Hence, the budget constraints are:

$$c_{t,0} + a_{t+1,1} = (1 - \tau_t)w_t$$

$$c_{t+1,1} = \frac{a_{t+1,1}(1 + r_{t+1})}{s_{t+1,1}} + d_{t+1}$$

The aggregate labor supply is equal to the number of young people in the economy: $L_t = N_{t,0}$. Assuming a Cobb-Douglas production function ($\rho = 1$), $Y_t = K_t^{\psi} (A_t N_{t,0})^{1-\psi}$ with $A_t = (g^A)^t A_0$ and depreciation of capital $0 < \delta < 1$, profit maximization of firms implies:

$$r_t + \delta = \psi k_t^{\psi - 1}$$
$$w_t = (1 - \psi) A_t k_t$$

¹¹This section follows closely section 2 of Krueger and Ludwig (2007).

where $k_t = K_t/(A_t N_{t,0})$ denotes the stock of capital per efficiency unit of labor.

The budget balance of the government running the PAYGO pension system implies the following social security transfers:

$$d_{t+1} = \tau_{t+1} w_{t+1} \frac{g_{t+1}^N}{s_{t+1,1}}$$

where $g_{t+1}^N = N_{t+1,0}/N_{t,0}$ denotes the growth rate of the young generation. Assuming a fixed replacement rate \bar{d} the pension transfers must also satisfy:

$$d_t = \bar{d}w_t(1-\tau_t)$$

Finally, clearing of the capital market and the law-of-motion of capital require: $K_{t+1} = N_{t,0}a_{t+1,1}$, $K_{t+1} = (1 - \delta)K_t + I_t$. Rewritten in terms of efficiency units of labor:

$$g_{t+1}^{N} A_{t+1} k_{t+1} = a_{t+1,1}$$

$$g^{A} g_{t+1}^{N} k_{t+1} = (1-\delta) k_{t} + i_{t}$$

BGP characterization. As shown in Appendix D, the equilibrium on a *balanced-growth path* (BGP) with constant effective capital stock ($k_t = k$ for all t) can be fully characterized by the following capital demand (by firms) and supply (by households):

$$\iota^{D}$$
 : $r = \frac{g^{A}g^{N} - (1 - \delta)}{\iota} - \delta$ (3.1)

$$\iota^{S} : r = \frac{\tau}{(1-\tau)\frac{\beta s_{1}}{g^{A}g^{N}} - \iota \frac{1+\beta s_{1}}{(1-\psi)[g^{A}g^{N}-(1-\delta)]}} - 1$$
(3.2)

where ι denotes the investment-output ratio which equals the aggregate saving rate (given the assumption of closed economy). Assuming a PAYGO pension system with fixed replacement rate \bar{d} the tax rate is given by:

$$\tau = \frac{\bar{d}}{\bar{d} + \frac{g^N}{s_1}} \tag{3.3}$$

Aging shock channels. Aging is characterized by decreasing fertility and increasing survival probabilities. In this simple model the former is represented by decreasing g_{t+1}^N , the latter by increasing $s_{t+1,1}$, both constant on a BGP. The BGP characterization above allows to determine the impact of aging in terms of movements of the capital demand (ι^D) and supply (ι^S) . Figure 1 depicts such movements assuming that the economy moves from an initial stationary equilibrium (BGP₀) to a final one (BGP_f) facing a reduction of $g^A g^N$ from 1.025 per annum to 1 and an increase of s_1 from 0.146 to 0.561 (according to the empirical old-dependency ratio, see Figure A.1).¹² It allows to detect four channels through which aging impacts the equilibrium real interest rate.¹³

¹²Notice that in the two-period model the BGP survival probability $s_1 = N_1/N_0$ is the old-dependency ratio, i.e. the number of elderly (aged 65+) over the number of workers (aged 15–64).

¹³Cf. Carvalho et al. (2016) in the context of a model à la Gertler (1999).



Figure 1: Two-period model: capital market equilibrium

Note. Movements of capital demand (ι^D) and supply (ι^S) to aging (i.e. decrease of g^N , increase of s_1). See equations (3.1), (3.2) and Appendix D for derivation. BGP₀ and BGP_f denote the initial and final balanced-growth path (BGP), respectively. It is assumed that one period corresponds to p = 30 years. The term g_f^N indicates that the unique demographic shock is in the labor growth, taking its assumed value in the final BGP $(g^A g^N = 1 \text{ compared to a value in the initial BGP of 1.025 per annum, i.e. <math>g^A g^N = 1.025^p$). From the initial to the final BGP s_1 goes from 0.146 to 0.561 (according to the old-dependency ratio, see Figure A.1). With an annual depreciation rate of $\delta_a = 0.0952$ (as in the subsequent section), the depreciation is $\delta = 1 - (1 - \delta_a)^p \approx 0.95$. The capital bias in production has the standard value of $\psi = 0.33$. In the initial BGP the target is an investment-output ratio $\iota = 0.2$ which implies an annual real interest rate of about $0.0419 = r = \{\psi[g^A g^N - (1 - \delta)]/\iota + (1 - \delta)\}^{1/p} - 1$. The replacement rate is set to $\overline{d} = 0.45$, implying a tax rate τ of about 6% in the initial BGP, 20% in the final BGP. These values imply an individual discount factor $\beta = \{\iota/[g^A g^N - (1 - \delta)] + \tau(1 - \psi)/(1 + r)\}/\{(1 - \tau)(1 - \psi)s_1/(g^A g^N) - \iota s_1/[g^A g^N - (1 - \delta)]\}$. The annualized effective discount factor: $(s_1\beta)^{1/p} = 0.9798$ in the initial BGP. The term τ_f in the legend indicates that the unique demographic shock is in the tax rate, taking its final BGP value.

- 1. Labor supply. A decrease of g^N induces firms to demand less capital (downward shift of ι^D) as it is optimal to reduce the capital stock in support of relatively scarcer labor input. At fixed capital supply, this decreases the real interest rate by making capital relatively more abundant than the labor employed in production, thus depressing the marginal product of capital all else being equal.
- 2. *Life expectancy*. An increase of the survival probability s_1 , all else being equal, makes households more patient (notice that it always multiplies the discount factor β) making them will-

ing to save more in the expectation of financing consumption for a longer life, for given retirement age, thus contributing to depress the real interest rate (shift to the right of ι^S).

- 3. Savers/dissavers composition. While a higher survival probability s_1 pushes up the saving rate, a decrease of the fertility growth g^N per se (in absence of a pension system) tends to decrease it, all else being equal. The reason is that at a lower g^N there are relatively less young people who are the only savers in the economy as compared to retirees. A lower saving rate makes capital scarcer, thus increasing the real interest rate (shift to the left of ι^S).¹⁴
- 4. *Pension system*. In presence of a PAYGO pension system with fixed replacement rate, aging increases the tax rate (see equation (3.3)). With less income-generating workers (lower g^N) in support of more old people (higher s_1) the fiscal authority needs to set a higher tax rate to have the budget balanced in each period and provide the agreed social security transfers. As the tax rate increases there is a crowding-out effect of private capital (shift to the left of ι^S). As capital becomes less abundant the equilibrium real interest rate increases when the tax rate increases, all else being equal. If the tax rate was always zero the savings by households would be higher.

Without any further assumption it is not possible to find a unique closed-form solution for the equilibrium real interest rate by equating the capital demand to the capital supply in (3.1) and (3.2). However, assuming full depreciation of capital ($\delta = 1$) is sufficient to have the following closed form solution:¹⁵

$$\iota = \frac{s_1 \beta \psi (1 - \psi) (1 - \tau)}{(1 + s_1 \beta) \psi + \tau (1 - \psi)}$$
$$r = -\frac{1}{\iota} g^A g^N - 1$$

In sum, this section reveals that in a stationary equilibrium such as a BGP with constant effective capital stock, aging tends to reduce the real interest rate via two main channels. On the one hand, aging makes the available labor in production relatively scarcer (decrease in g^N , equivalent to a decrease of labor-augmenting productivity growth g^A) inducing firms to demand less capital (channel 1); on the other hand, aging makes individuals more willing to supply capital (increase in s_1) thus increasing the investment-output ratio (ι) – which in a closed economy equals the gross

$$\iota = \frac{\beta s_1}{1+\beta s_1}(1-\psi)\left(1-\frac{1-\delta}{g^Ag^N}\right)$$

¹⁴Notice that in absence of a pension system, $\tau = 0$, the savings supply becomes inelastic. From equation (3.2), for 1 + r > 0, it results:

As g^N decreases, ι decreases (i.e. ι^S shifts to the left) via channel 3. However, this impact tends to be small as δ is not far from 1 over a period (corresponding to 30 years) in the model. In case of $\delta = 1$, channel 3 is nil in this setting. ¹⁵Notice that since a period here corresponds to 30 years, it is reasonable to assume full depreciation.

saving rate (channel 2). Both channels independently decrease the real interest rate while the latter can be dampened by the crowding out effect on savings associated with the increase of the tax rate implied by a PAYGO pension system with defined benefits (channel 4). Therefore, a stationary equilibrium with a higher survival probability is unambiguously characterized by a higher investment-output ratio. The potential downward impact on the saving rate (hence upward impact on the real interest rate) stemming from the increasing number of dissavers does not materialize in a stationary equilibrium given the current setting. This of course does not exclude that along the transition dynamics from an initial to a final stationary equilibrium the saving rate can decrease while the number of dissavers increases (channel 3).¹⁶

4 Quantitative analysis

The goal of the quantitative analysis is to study the transition dynamics of the macroeconomic system of section 2 from an initial to a final steady state, where the unique perfectly-anticipated exogenous driving process is the time-varying demographic structure. The focus is on the euro area composed by 12 countries (EA12 henceforth) modeled as a closed economy.¹⁷

4.1 Experiment and baseline calibration

One period of the model corresponds to one year. Following Domeij and Flodén (2006), equation (2.1) is directly used to retrieve the conditional survival probabilities using data on the empirical number of people, $N_{t,j}$, by single age-group j for each year t in the time-range 1950-2100. Data are taken from the United Nations (UN, 2017) World Population Prospects: The 2017 Revision including the medium variant projections until year 2100.¹⁸ Hence, the demographic evolution in the model is fully captured by the incoming cohort size ($N_{t,0}$) and the implied unconditional survival probabilities $\pi_{t+j,j}$ for all t and j via the recursive formulation: $N_{t+j,j} = \pi_{t+j,j}N_{t,0}$.¹⁹

It is assumed that in the initial steady state the system has the demographics prevailing in year 1951 in the data, i.e. for all j = 0, 1, ..., J - 1 the demographic structure is given by: $N_{1951,j+1} = \pi_{1951,j+1}N_{1951,0}$. The experiment is such that while in 1950 (as well as in all previous periods) the

¹⁶Only one curve shift in Figure 1 has not been discussed above. That pertains to the shift of ι^S with $\tau > 0$ when g^N decreases (" ι^S, g_f^N " line). The saving supply shifts slightly to the right. The reason is that with a lower g^N for given τ the pension transfers decrease. Hence, with the goal of smoothing consumption, individuals want to save more. The quantitative importance of this channel will be discussed later when comparing a PAYGO pension system with fixed vs endogenous replacement rate.

¹⁷EA12 is composed by the following countries: Austria (AT), Belgium (BE), Finland (FI), France (FR), Germany (DE), Greece (EL), Ireland (IE), Italy (IT), Luxembourg (LU), Netherlands (NL), Portugal (PT), Spain (ES).

¹⁸Before year 1990, the number of people aged more than 80 are grouped together in the set 80+ for all countries. As a strategy to identify the number of people in each single age group after age 80 for years 1950-1989, the implied survival probabilities of 1990 for those aged more than 80 have been applied backwards. The shape of the number of people of different ages in each year has been slightly smoothed via the Matlab "smooth" function with quadratic fit and a span of 0.08.

¹⁹Figure A.2 plots both $N_{t,0}$ and $\pi_{t+j,j}$ for all years t from 1951 to 2100 and all ages in the model.

system is assumed to be in the initial steady state with the 1951 demographics, in 1951 there is the information shock: agents learn about the new demographic development for all subsequent years, and what this implies for macroeconomic variables in a perfect-foresight environment. In year 2100 the conditional survival probabilities and the incoming cohort size start remaining fixed forever. This implies an evolution of the demographic structure that eventually gets stationary again so that around year 2200 the system reaches a new final steady state.²⁰

Given the demographics, the structural model parameters and the implied solution values for the initial and final steady state, the dynamic equilibrium is solved using a standard deterministic simulation set-up where the numerical problem of solving a nonlinear system of simultaneous equations is managed by means of a Newton-type method.²¹

The baseline calibration employs standard parameter values summarized in Table 1. The role of most parameter values will be explored in subsequent sections via sensitivity analyses, including an exercise where the model is fed with a series measuring total-factor-productivity growth in the data to capture technology growth. Here, to isolate the effect of demographics, the labor-augmenting technological parameter A_t is simply set to 1 (without loss of generality) in all periods t. Therefore, the model in the baseline features no technology growth.²²

In the baseline calibration the representative firm employs a Cobb-Douglas production function which corresponds to the case when the elasticity of substitution between the factors of production tends to 1. Consistently with a widely used value in the literature the bias towards capital in the production function ψ is set to about 1/3.

The constant relative risk aversion (CRRA) parameter σ , which is the inverse of the intertemporal elasticity of substitution, is set to the standard value of 1, a widely used assumption in the literature (see e.g. Krueger and Ludwig (2007)), implying that the representative households have logarithmic preferences over consumption.

The fixed net replacement rate \bar{d} of the PAYGO pension system is set 0.45, in line with what assumed by Kara and von Thadden (2016) in a model for the euro area.²³

²⁰The dynamic simulations of the model are run until year 2351.

²¹This method is employed under the "perfect foresight solver" available in *Dynare*. Specifically, the Jacobian of the system has dimension 177×400 given that there are 177 endogenous variables over 400 periods (years). On the households side there are 85 endogenous variables for consumption and as many minus 1 for assets holdings (the initial asset is exogenously given, equal to zero) for a total of 169. The remaining endogenous variables are: real interest rate, wage rate, tax rate, pension transfers, and aggregate capital, assets, income, consumption. In the first 150 periods (from 1951 to 2100) the system has a different demographic structure in each period as in the data by UN (2017). In the remaining periods, a time-span more than sufficient for the system to reach the final steady state, the conditional survival probabilities and incoming cohort sizes take the values of 2100.

²²This is standard practice with the argument of isolating the effect of demographic factors. See e.g. De la Croix et al. (2013), Gagnon et al. (2016).

²³This number is roughly the average across the EA12 countries of the replacement rates reported by Barany et al. (2018) who use the same type of PAYGO pension system adopted in this paper to compare most of the world countries. Their measure is based on an upgrade of the official replacement rates to take into account concerns about measurement errors or potential biases in the percentages of retirees receiving benefits and working-age population contributing to pensions.

Parameter	Value	Note			
Structural:					
A_t	$1 \ \forall t$	labor-augmenting technology, no technology growth			
ho	1	capital-labor substitutability, Cobb-Douglas production function			
	0.33	bias towards capital in the production function			
δ	0.0952	depreciation rate of capital			
σ	1	constant relative risk aversion (CRRA), log preferences			
\overline{d} 0.45		net replacement rate of PAYGO pension system			
β	0.9598	individual discount factor, endogenous in the initial steady state: $K/Y = 2.1$			
Demographic:					
J	84	terminal life-age (99)			
j_r	50	retirement age (65)			
h_j	Figure A.3	individual life-cycle effective labor supply. Source: Domeij and Flodén (2006)			
$N_{t,0}$	Figure A.2	incoming cohort size (age 15). Source: UN (2017)			
$\pi_{t+j,j}$	Figure A.2	unconditional survival probabilities, retrieved from UN (2017) applying (2.1)			

Table 1: Baseline calibration: parameter values

For what concerns demographics, individual labor supply in efficiency units, h_j , is interpolated using the data-points provided by Domeij and Flodén (2006), obtained by interacting the profiles of both productivity and the participation rates.²⁴ It is assumed that households enter the world as workers at age 15, all retiring at age 65 which correspond to $j_r = 50$ (this is why h_j drops abruptly to zero after age 64). A standard assumption, see Kara and von Thadden (2016), Bielecki et al. (2018) for the euro area, in line with what reported for most OECD economies (see Carvalho et al. (2016), Table 2). Households live at most until age 99 corresponding to J = 84 so that in each year there are 85 overlapping generations.

The initial steady state is calibrated to a capital-output ratio of 2.1 and an investment-output ratio of 0.2. These values are such that the capital-output ratio and the investment-output ratio generated by the simulation are in line with what can be found in the data using series for the gross fixed capital formation,²⁵ and in line with values normally found in the literature.²⁶ Furthermore,

²⁴See Figure A.3. A similar profile is employed for European countries by Cooley et al. (2019), see their Figure 7. Following a common assumption in OLG modeling (that can be found in most of the literature mentioned in this paper), labor force participation rates are assumed to be constant over time. Hence, the model is not tailored to directly capture structural changes such as the increase in female labor force participation. Attanasio et al. (2007) explicitly take into account the time-varying nature of female participation rates and find that the "participation transition is virtually exhausted in 2005 in the North" (where the North includes: North America, Europe, Japan, Australia, and New Zealand). Hence, the impact of participation on factor prices should be limited after 2005. This is reassuring given that, as shown later, most of the downward impact of demographic change on the real interest rate is found to occur after the mid-2000s. Furthermore, extreme variations in h_j (such as those studied in section 4.4.3) are not found to have such a comparably sizable effect on the real interest rate.

²⁵ To have empirical estimates for the capital-output ratio, the series used here are: "Gross fixed capital formation (constant LCU)" and "GDP (constant LCU)", data source: World Development Indicators (WDI) by the World Bank (update: January 2018). The capital stock is estimated by applying the perpetual inventory method (cf. e.g. Caselli (2005)). The initial capital stock (the base year is 1970, the first year data are available for all EA12 countries) is computed using the formula: $K_{1970} = I_{1970}/(g_I + \delta_K)$ where I_{1970} corresponds to the gross capital formation in 1970, g_I is the average growth rate, while δ_K is set to 7.5%. The capital stock is obtained via the law-of-motion: $K_{t+1} = (1 - \delta)K_t + I_t$ for all years t from 1970 to 2016. The series for EA12 are obtained by weighting each country with its real GDP share in year 2000.

²⁶For example, De la Croix et al. (2013) target a capital-output ratio of 2.4 in 2010 for France; Attanasio et al. (2007)

they give an annual depreciation rate of capital $\delta = 0.0952$ which is consistent with what set e.g. by Gomes et al. (2012) in a model for the euro area.²⁷ Hence in the initial steady state the real interest rate is about 6.19% per annum.²⁸ While the reader might find this value relatively high, especially if compared with recent work on the topic, it is in line with seminal contributions in the literature.²⁹

Given these targets and structural parameter values the annual individual discount factor β is solved endogenously (by means of a solver for non-linear systems of equations) in the initial steady state. The resulting value is 0.9598.



Figure 2: Demographic growth rates in EA12

Note. The series plotted are the annual growth rates of population $N_t = \sum_j N_{t,j}$, effective labor supply $L_t = \sum_j h_j N_{t,j}$ (see Figure A.3 for the age-dependent units of efficiency h_j), and their ratio, where the ages j considered involve people aged between 15 and 100, consistently with the model. The number of people $N_{t,j}$ is taken from data provided by the United Nations (UN, 2017) *World Population Prospects: The 2017 Revision* for year $t \in [1950, 2100]$, medium variant after year 2016 (see footnote 18).

Figure 2 plots the annual growth rates of population $N_t = \sum_j N_{t,j}$, aggregate effective labor supply $L_t = \sum_j h_j N_{t,j}$, and their ratio. The growth rate of the effective labor to population ratio

have a value of 2.5 in 2005 for a composite of advanced economies representing the "North"; Domeij and Flodén (2006) have a slightly higher number of 2.74 for the world in 1990. Compare these numbers with Figure 4 below.

²⁷Notice that targeting a capital-output ratio K/Y = 2.1 with an investment-output ratio I/Y = 0.2 gives, by the lawof-motion of capital in steady state a depreciation rate $\delta = (I/Y)/(K/Y) = 0.2/2.1 \approx 0.0952$. Gomes et al. (2012) set the depreciation rate at the quarterly frequency to 0.025. At the annual frequency this implies a depreciation rate $1 - (1 - .025)^4 \approx 0.0963$.

²⁸From the representative firm first order condition with respect to capital the real interest rate is $\psi/(K/Y) - \delta = 0.33/2.1 - 0.0952 \approx 0.0619$.

²⁹See e.g. Domeij and Flodén (2006) (world), Attanasio et al. (2007) ("North" = "more developed regions" according to UN), Krueger and Ludwig (2007) (world), De la Croix et al. (2013) (France). They all target a real interest rate in the range of 6–7% per annum in a period spanning from 1950 to 2010. Among the recent contributions with a lower real interest rate target see e.g. Eggertsson et al. (2019) (US), Bielecki et al. (2018) (EU), Gagnon et al. (2016) (US). Section 5 will explore a different calibration in the attempt to match data and econometric estimates for the euro area from different sources. Controversy on the point arises because the model, as well as the models in the literature mentioned above, features a unique asset; and because the natural real interest rate is an unobserved variable.

is a key determinant of the aggregate dynamics of the system along the transition path. Between 1985 and 2030 it decreases by about 1 percentage point going from a growth rate of 0.2% to -0.8% per year. While the growth rate of both population and effective labor decrease over time until 2030, the latter decreases more turning negative after 2010 which is approximately the year when the baby boom generations start to retire (at age 65). The retirement of those generations determine a widening of the gap between the population and labor force growth rates until 2030. After that there is a re-bouncing of the growth rate of the effective labor to population ratio as the increasing scarcity of effective labor becomes less strong after the baby boom generations have ceased to exist. But the growth rate of population and effective labor are both negative in the projected horizon. In the period between 1990 to 2030 the scarcity of effective labor is exacerbated by the fact that the demographic distribution is tilted towards relatively older people who are calibrated to provide relatively less effective labor.

Once the medium variant projections from UN (2017) are used until 2100, the model predicts that the real interest rate decreases in the long-run (while the real wage increases) going from 6.19% in the initial steady state to 4.56% per annum in the final steady state. It results that individual labor (pension) income is slightly higher (lower) in the long-run. Furthermore, as factor prices adjust in the long-run saving becomes less attractive (and borrowing during the early stages of life more attractive) so that the individual age-profile of asset holdings in the final steady state lies always below its equivalent in the initial steady state. However, as the population distribution shifts in the long-run in favor of relatively older people, the distribution of both aggregate savings and aggregate consumption in the long run has relatively more mass on older ages.³⁰

Figure 3 compares the life-cycle profiles generated by the simulated model for a selection of generations (continuous black lines) along the transition dynamics with the data for the European countries available in the *National Transfer Accounts* (see note under Figure 3). The model finds visual correspondence with the data. Few features are worth emphasizing.

First, the calibration of the individual labor supply in efficiency units, h_j , with the assumption of exogenous retirement at age 65, allows to match the hump-shaped labor income profile found in the data, roughly corresponding to the median across the labor income profiles for the countries observed.

Second, the hump-shaped profile of labor income coupled with the consumption-smoothing motive entails that households start to build up positive savings only after the first 20-25 years of working life (when they are 35-40 years old). This is a standard life-cycle profile captured by OLG models (cf. e.g. Cooley and Henriksen (2018)) and finds some correspondence in the data.³¹

³⁰These patterns are confirmed in Figure A.4 showing not only the life-cycle profiles for the initial and the final steady state but also counter-factual (partial equilibrium) profiles for the final steady state where the real interest rate is fixed at the initial steady state value (see dotted lines). It confirms that had the system not incurred any general equilibrium adjustment in factor prices, households – facing higher survival probabilities in the long-run – would have saved more in all life-cycle periods in the long-run.

³¹Compared to data for consumption and labor income, data for total savings shown in Figure 3 are available for less countries, seem to be measured with more noise and suggest more variability across countries.

Third, the individual consumption profile tends to be upward sloping both in the model and in the data. In the model this feature is due to the assumption of a "perfect annuity market" which ensures that there exists a perfect insurance against individual lifetime uncertainty as the end-of-period assets of the households prematurely dying are equally redistributed among the surviving members of the same generation (see footnote 8). Practically, this assumption ensures that the conditional survival probabilities do not enter the inter-temporal conditions (see (B.1) in Appendix B) preventing a hump-shaped profile of consumption in steady state.³² As endogenous forces lead the real interest rate to decrease progressively over time (as shown in the next sections) the consumption profile of subsequent generations tends to be flatter.³³



Figure 3: Profiles of individual consumption, labor income and savings: model vs data, normalized on mean for persons 50-60 years old.

Note. The grey markers with no line represent data from the *National Transfers Accounts* www.ntaccounts.org (NTA), a global project whose main reference is Lee and Mason (2011). The European countries for which data is available are (year used in parenthesis - often only one year is available, results are not sensitive to the specific year used): Austria (2005), Finland (2004), France (2005), Germany (2003), Italy (2008), Spain (2000). For each age of the life-cycle until 90+ (or 100 in few cases) the dataset includes data for the per capita ("smooth mean") components of the following identity: consumption = labor_income + asset_income - saving + transfers which finds correspondence in the model. The NTA report the flows for a particular year for a cross-section of age groups. Hence the patterns in the data reflect the effects of age as well as cohort differences. To facilitate comparison both the data and the model series are normalized to the respective mean for persons 50–60 years old. Data on saving available only for Austria, Finland, Italy, Spain. The *continuous black line* represent the life-cycle profiles in the model along the transition dynamics for a selection of generations, those born (aged 15) in: 1971, 1981, 1991, 2001.

 $^{^{32}}$ See Appendix H for a relaxation of the perfect annuity market assumption.

³³See Lee and Mason (2011) on how the empirical results based on the *National Transfer Accounts* might differ from those based on e.g. the US *Consumer Expenditure Survey* where the life-cycle profile of consumption is generally found to be more hump-shaped (cf. Fernndez-Villaverde and Krueger (2007)).

4.2 Baseline results

Figure 4 shows the resulting transition dynamics under the baseline calibration (see Table 4.1) for the main macroeconomic variables with (solid line) and without (dashed line) PAYGO pension system. Demographic change is the only exogenous driver in the model.

A decrease of the growth rate of the effective labor to population ratio is akin to a negative shock in the growth rate of total factor productivity (TFP) for output per capita growth. This is why the annual growth rate of output per capita follows closely the growth rate of the effective labor to population ratio.³⁴ Similarly, the growth rate of total output follows closely the growth rate of effective labor. Compare the first panel of Figure 4 with Figure 2. While the presence of the PAYGO pension system affects the level of output by crowding-out capital accumulation, the growth rate of output is essentially unaffected even though along the transition it stands always at a slightly higher level when the PAYGO pension system is not in place. Instead, by assumption, the growth rate of output in both the initial and final steady state is zero.

As emphasized above, aging means lower fertility rates and higher survival probabilities. As (effective) labor supply becomes increasingly scarcer over time and households are willing to save more in the expectation of a longer life (for given retirement age), capital becomes relatively more abundant than labor in the economy. Hence, the capital-labor ratio (and so the capital-output ratio) increases over time, depressing the marginal product of capital and so the equilibrium real interest rate. This is the most intuitive impact of aging, encompassing channel 1 and 2 above (section 3), and seems immediately clear in the acutest phase of the aging process, between 1990 and 2030.

Notice, however, that in this period the aggregate saving rate (equal to the investment rate by the assumption of closed-economy) decreases. In principle, as shown in the simple model of section 3, there can exist a negative relationship between the saving rate and the real interest. But here they both decrease. What is the intuition? The saving rate starts decreasing few years before 2010, roughly when the first wave of baby boomers start retiring, and keeps doing so until about 2040, when most of the baby boomers have most likely chased to exist. It goes from about 23% of output in 2010 to 21% in 2040. It is conceivable that part of the decrease of the saving rate is due to the flow of dissaving that originates from the retiring baby boomers (channel 3 above). Additionally, one needs to consider not only that with labor becoming relatively scarcer it is optimal to reduce the capital stock to exploit the complementarity between factors in production, but also other general equilibrium effects that connect the decrease of the real interest rate to the saving rate.

Nonetheless, the capital stock in the economy decreases slowly over time and so does the holding of wealth over the course of retirement. Since it is the capital stock rather than the flow of saving that determines the real interest rate, the general equilibrium analysis under the baseline calibration

³⁴To see this, consider the Cobb-Douglas production function: $Y_t = K_t^{\psi} L_t^{1-\psi}$. Divide by the total number of people in the economy, N_t , to have variables per capita, denoted by an over-line. Then, $\overline{Y}_t = \overline{K}_t^{\psi} \overline{L}_t^{1-\psi}$. Hence the growth rate of output per capita: $\overline{Y}_{t+1}/\overline{Y}_t = (\overline{L}_{t+1}/\overline{L}_t)^{1-\psi} (\overline{K}_{t+1}/\overline{K}_t)^{\psi}$. That is, a decrease of the growth rate of effective labor to population ratio, $\overline{L}_{t+1}/\overline{L}_t = (L_{t+1}/N_{t+1})/(L_t/N_t)$, is akin to a negative shock to TFP growth.

shows that ultimately the main driver in the period between 1990 and 2030 is the relative scarcity of labor (channel 1) coupled with an increased willingness to save by households discounting higher survival probabilities (channel 2) in spite of general equilibrium forces that lead the saving rate to decrease. As a matter of fact, the real interest rate starts re-bouncing around year 2030, when the growth rate of the effective labor to population ratio start re-bouncing too (see Figure 2). However, in spite of the re-bouncing, in the very long-run the real interest rate is permanently at a lower level compared to its initial steady state value as households discount permanently higher survival probabilities.



Figure 4: Baseline transition dynamics in EA12

Note. Calibration of Table 1. In the "no paygo" scenario the tax rate is $\tau_t = 0$ for all years t and the individual discount factor β is recalibrated to match the target capital-output ratio in the initial steady state. The aggregate saving rate in the figure is: $I_t/Y_t = [K_{t+1} - (1 - \delta)K_t]/Y_t$.

The dashed lines in Figure 4 show what happens when the PAYGO pension system is not in place, i.e. when households need to finance the retirement period by means of private savings only. It is immediately clear that without social security benefits there is no crowding-out effect on capital so that both the saving rate and the capital-output ratio are always at a higher level, going

hand in hand with a lower real interest rate along all the transition (channel 4 above). There is not only a level effect but also a relative magnitude effect: between 1990 and 2030 the real interest rate decreases by 1.3 (1.56) percentage points with (without) the PAYGO pension system in the economy. If the economy did not run any pension system, the model predicts that the real interest rate by 2030 would stand at a lower level of about 0.5 percentage points (the real interest rate in 2030 is about 4.1% with PAYGO system, 3.6% without).

Therefore, while it is true that the presence of a PAYGO pension system mitigates the downward impact of aging on the real interest rate (by crowding-out capital investment, a well established result in the literature), under the baseline calibration it does not seem the case that demographics can meaningfully "reverse three multi-decade global trends" on real interest rates via this channel as argued by Goodhart and Pradhan (2017). The next sections will explore whether this might or might not happen studying the transition dynamics under different calibrations and assumptions.

Finally, it is interesting to notice that before 2007 one could easily interpret the impact of demographic change on output growth as a type of scaling factor, producing if anything mild variations around a constant, close to zero for the case of output per capita. However, after 2007 – which, coincidentally, is when the "global financial crisis" started - the model predicts a dramatic decrease of output growth due to demographic change.³⁵ While in 2007 demographics was contributing to increase total output by 0.4% per year, by 2015 the figure is -0.2%. Going to 2026 the model predicts a further drag on total output with an annual growth rate of about -0.6%. A similar persistent decreases is predicted for output per capita annual growth rate: it goes from -0.2% in 2007 to -0.4% in 2015, to -0.7% by 2026. For the real interest rate too, the most dramatic phase is after 2007 with a variation of about -1 percentage point between 2007 and 2030 (compared to -1.3 percentage points between 1990 and 2030). Hence, what noticed by Gagnon et al. (2018) for the United States seems also true for Europe: "the largest effects of demographics on interest rates and GDP growth coincide to some degree with the Global Financial Crisis". So that "downward pressures on interest rates and GDP growth due to demographics could be easily misinterpreted as persistent but ultimately temporary influences of the global financial crisis" (Gagnon et al., 2016). Instead, what OLG models seem to predict is a "new normal".³⁶

³⁵This is made clearer in Figure A.5 (that also reports the results of different simulations varying the intertemporal elasticity of substitution σ , see section 4.4.1).

³⁶For completeness, Figure A.6 reports the transition dynamics for output, consumption and capital per capita as well as for the real interest rates until year 2200 when the system has essentially reached the final steady state after the demographic evolution has become stationary again once the incoming cohort size and the conditional survival probabilities have been fixed at the 2100 level forever. For the way the experiment is designed, consumption decreases upon impact in 1951 while both capital and output are fixed (and so the real interest rate) as the former is a pre-determined variable and the latter depends both on capital and labor which, by assumption, has the same value of 1950. This downward jump of aggregate consumption determines an upward jump of the aggregate saving rate that goes from 20% of the initial steady state to slightly more than 23% in 1951. See Figure 4. Figure A.7 explores the jumps of consumption in 1951 on the age-profiles for different cohorts. In 1951 households have to re-optimize given the new set of information about the future. Most of the generations respond to higher survival probabilities in the future by saving more (i.e. with a consumption profile that tends to jump downward upon impact). However, younger generations (roughly younger than 30 in 1951) have *ex-post* saved too much during the first years of working

4.3 Isolating the drivers

Figure 5 shows the results of running different counterfactual simulations where the conditional survival probabilities are fixed at their level in the first period of the transition dynamics;³⁷ and where the economy is assumed not to be endowed with a PAYGO pension system (in this case the individual discount factor β is recalibrated to match the target capital-output ratio in the initial steady state, as in Figure 4). The reference year is chosen to be 1987 as in this year the baseline simulated series takes a value, normalized to zero in Figure 5, approximately equal to its mean over the 1980s.

Notice that (see equation (2.1)) the evolution of the population distribution in the model depends on three elements: the initial population distribution, the incoming cohort size (the newborns in the model, aged 15) and the conditional survival probabilities. When the conditional survival probabilities are fixed at the first period level, the model counterfactually captures not only the impact on savings decisions (channel 2) but also the shift of the population towards older cohorts (which tend to be dissavers, channel 3) due to higher survival probabilities. Furthermore, the labor supply in this case follows an implied evolution that does not only depend on the incoming cohort size and the initial population distribution but also on the fixed survival probabilities themselves. In this sense, fixing the survival probabilities isolates only imperfectly the life expectancy, the labor supply and the composition channels above. The residual difference between the "fixed survival probabilities" line and the zero line can be attributed to forces other than those due to the increase in the survival probabilities. In particular, from equation (2.1), the only exogenous variation in this case is given by the incoming cohort size in each period, thus capturing the effects of variations in fertility rates over time.

Through the lens of this decomposition, the only force that provides upward pressure to the real interest rate is the presence of the PAYGO pension system (channel 4) but, as already seen in the previous section, it is not strong enough to counteract the other forces. Compared to the mean in the 1980s the real interest rate decreases about 1.4 percentage points until 2030. If no PAYGO pension system was ever run the real interest rate would be about 0.5 percentage points lower by 2030. The increase of the survival probabilities over time has an important role in explaining the decrease of the real interest rate in the model. Alone they are associated with a negative deviation of the real interest rate from the mean in the 1980s of about 0.9 percentage points by 2030 (cf. the "Mortality" shaded area in Figure 5). The residual decrease of about 0.5 percentage points by 2030 can be attributed to forces related to the decrease of the newborns in the model (cf. the "Fertility"

⁽when they were planning for a steeply upward-sloping consumption profile based on higher real interest rate). Now, learning not only about higher survival probabilities but also lower real interest rate in the future, they re-optimize tending to save less for the future (correcting in turn the 'excessive' capital accumulation in the past).

³⁷The first period of the transition was preferred to the initial steady state to fix the survival probabilities at. The reason is that the initial steady state does not use actual survival probabilities, for the way the model is built. It uses instead the ones implied by the demographic structure prevailing in the chosen year (1951). Consequently, the counterfactual results so obtained are better interpreted in terms of a reference year standing some time later than the initial steady state, to avoid also potential front-loading effects, as done in Figure 5.



Figure 5: Transition dynamics in EA12: different channels

Note. Calibration of Table 1. The results are obtained by running the simulations fixing the conditional survival probabilities at the first period of the transition dynamics and assuming that the economy is not endowed with a PAYGO pension system (cf. dashed line in Figure 4). The reference year is 1987 (when the baseline series takes approximately its mean value over the 1980s, normalized to zero).

shaded area in Figure 5). These findings are in line with the independent work by Bielecki et al. (2020) where, in a similar OLG model applied to the euro area, they find: "Of the two demographic factors, mortality is the more important one. Fertility also plays a significant role, but only since about 2000, and its contribution is even slightly positive in the 1990s because of the echo effect of the postwar baby boom".³⁸

Interestingly, the fertility headwind fades away progressively after 2030 (at its trough) until it is essentially reabsorbed by 2050. The reason is that in the projected horizon the growth rate of the newborns in the model gets progressively less negative until reaching again zero, as in the initial steady state. This is the main driver of the re-bouncing after the 2030. Hence all the permanent decrease of the real interest rate in the very long run, about 1 percentage point compared to the average in the 1980s, is due to the permanent increase in the survival probabilities. This should not be surprising in light of the discussion in section 3. As the growth rates of the incoming cohort size and technology are set to zero in the (initial and final) stationary equilibria, all the demographic effects on the real interest rate in the long run depend on the survival probabilities. Section 5,

³⁸The relative importance of the mortality channel is generally found in similar OLG models even for other countries. For example, Krueger and Ludwig (2007) report: "[...] the main sources of changes in factor prices and other aggregates between 2005 and 2080, at least in the U.S., come from changes in mortality rates". Also the results in Carvalho et al. (2016) and Lisack et al. (2017), for a representative OECD economy, point to the relative importance of the life expectancy channel. Gagnon et al. (2016) instead, in a OLG model for the US, find a more limited role for mortality changes, at least from 1980 to 2015.

particularly subsection 5.3, will explore these results in presence of non-zero technology growth.

4.4 Sensitivity analysis

The goal of the sensitivity analysis is to understand how the results change when the model departs from the baseline calibration. This sheds light not only on the robustness of the quantitative estimates but also on the economic significance of the main parameters in the model.

4.4.1 Intertemporal elasticity of substitution

In the current setting, the intertemporal elasticity of substitution in consumption is represented by the inverse of the constant relative risk aversion (CRRA) coefficient, σ . By changing the intertemporal condition for consumption, a variation of σ modifies the reaction of households to the higher survival probabilities brought about by aging, i.e. it modifies the transmission of channel 2 above. In particular, a higher σ , by making the consumption-smoothing motive stronger, would tend to amplify channel 2, thus tending to reduce the real interest rate even more than in the baseline.

Eichenbaum et al. (1988) suggest that an appropriate range is $\sigma \in [0.5, 3]$, a common range in the business-cycle literature. Rarely $\sigma < 1$. Most often σ is simply set to 1, which corresponds to the case of logarithmic preferences. In a systematic analysis of the literature on the estimates of the intertemporal elasticity of substitution, Havranek (2013) documents that "the typical range of calibrations lies between 0.2 and 2", corresponding to $\sigma \in [0.5, 5]$. However, the value of σ remains controversial.³⁹

In the literature on OLG models it is quite common to set σ either equal to 1 or to 2. Notable examples of quantitative OLG models calibrated for both Europe and the US, as well as other OECD economies, include with $\sigma = 1$: Krueger and Ludwig (2007), Gagnon et al. (2016), with $\sigma = 2$: Domeij and Flodén (2006), Attanasio et al. (2007).

Figure 6 shows results of the transition dynamics under three different values of σ : 1 (baseline), 2, 2.5 with and without the PAYGO pension system. In each simulation the value of the individual discount factor β is recalibrated in order to match the target capital-output ratio of 2.1 in the initial steady state. Table 2 reports, for each experiment, the values of β as well as the percentage points change of the real interest rate between 1990 and 2030.

As σ increases the impact of aging on the real interest rate becomes more negative. For example, when $\sigma = 2$ the real interest rate decreases by 2.32 percentage points between 1990 and 2030 compared to a baseline decrease of 1.3 percentage points. Looking at Figure 6, to higher values of σ correspond paths of the real interest rate that lie at lower and lower levels in all periods of the transition. While the path of the real interest rate for the cases of no PAYGO pension system seems unrealistically low, it might serve as a useful counterfactual (for cases of σ not believed to

³⁹Havranek (2013) finds "strong publication bias: researchers report negative and insignificant estimates less often than they should, which pulls the mean estimate [of the inverse of σ] up by about 0.5", pointing that the "corrected mean of micro estimates ... is around 0.3–0.4", i.e. a mean value of σ between 2.5 and 3.34.



Figure 6: Transition dynamics in EA12: different values of σ

Note. Calibration of Table 1 with different values of σ . In each simulation β is recalibrated to be consistent with the target capital-output ratio of 2.1 in the initial steady state.

be unrealistically high) supporting the view that absent policy interventions (such as the PAYGO system) real interest rates would be at even lower levels than the those observed in the data.⁴⁰

It is interesting to note that for the cases of $\sigma = 2, 2.5$, especially for the no PAYGO scenario, the income effect seems to prevail over the substitution effect in general equilibrium so that the saving rate increases as the real interest rate decreases. With more accumulation of capital, a higher σ is also associated with a significantly higher growth rate of output along the transition.⁴¹

Overall, the big (probably unrealistic) impact that the model predicts for the values of σ higher than in the baseline hints to stay on the conservative side when calibrating the intertemporal elasticity of substitution in a frictionless modeling context where intertemporal considerations linked to demographic change are the only drivers of savings.

⁴⁰Cf. Rachel and Summers (2019a): "[...] neutral real interest rates would have declined by far more than what has been observed in the industrial world and would in all likelihood be significantly negative but for offsetting fiscal policies over the last generation".

⁴¹This is made clearer by Figure A.5.

	σ	β	$\Delta r_{1990-2030}$
	1	0.9598	-1.30
PAYGO	2	0.9783	-2.32
	2.5	0.9877	-2.71
	1	0.9566	-1.56
NO PAYGO	2	0.9717	-3.16
	2.5	0.9794	-3.63

Table 2: Different values of σ : impact on the real interest rate, 1990-2030 percentage points change

4.4.2 Capital-labor substitutability

As seen above, a driver of the declining real interest rate over the demographic transition after 1990 is the scarcity of labor as production input. As labor becomes scarcer, the demand for capital by firms is reduced thus leading to a decrease of the real interest rate everything else being equal (see channel 1 in section 3). A criticism to the quantitative estimates above based on this channel is that the demand for capital comes from the assumption of a Cobb-Douglas production function, i.e. a unitary elasticity of substitution between labor and capital ρ .

This channel might not matter at all if one admits that capital and labor are sufficiently substitutable between each other. In the limit, when capital and labor are perfect substitutes, changes in relative factor quantities have no impact on relative factor prices. To what extent can a higher substitutability between labor and capital mitigate the negative impact on the real interest rate induced by the demographic transition? Consider the CES production function in expression (2.7). When $\rho \rightarrow 1$, the production function is a Cobb-Douglas. When $\rho > 1$, capital and labor are said to be gross substitutes. In this case, a lower supply of one input leads to added demand for the other input. The opposite occurs when $\rho \leq 1$, in which case capital and labor are said to be gross complements. Therefore, when the labor supply decreases in the process of population aging with labor and capital as gross substitutes in production, the capital demand is affected differently compared to the case of a Cobb-Douglas production function.

In his summary of the empirical literature, Chirinko (2008) concludes that "the weight of the evidence suggests that ρ lies in the range between 0.40 and 0.60".⁴² Nonetheless, it might be interesting to consider values of ρ grater than unity. One could speculate that ongoing economic processes are changing the nature of capital, so that estimates based on historical data might not be reliable any longer. For example, automation can be thought as a process that by making labor increasingly superfluous in production is leading to an increasing degree of substitutability between labor and capital.⁴³ Furthermore, there might be reasons to believe that $\rho > 1$. Karabarbounis and

⁴²Of the 31 sources listed, 26 shows a ρ strictly less than one with a median of 0.52. The maximum value, reported only by one source, points to a value of 2. The remaining sources point to $1 < \rho < 1.5$ with one exception where ρ is slightly bigger than 1.5.

⁴³Acemoglu and Restrepo (2018) "argue theoretically and document empirically that aging leads to greater (industrial) automation, and in particular, to more intensive use and development of robots".

Neiman (2014) need ρ greater than one to explain the simultaneous decrease of the relative price of investment goods and the labor share of income. They estimate $\rho = 1.25$. Piketty (2014) too needs ρ greater than one to explain the fact that historically the capital share of income was lower when the capital-output ratio was lower, even though he does not provide estimates of ρ .⁴⁴



Figure 7: Transition dynamics in EA12: different values of ρ

Note. Calibration of Table 1 with different values of capital-labor elasticity of substitution ρ . For $\rho > 1(< 1)$ capital and labor are gross substitutes (complements). In each simulation the capital-output ratio in the initial steady state is allowed to differ from the baseline target value.

Figure 7 plots the real interest rate resulting from the demographic transition for three values of ρ : 1 (baseline), 0.5, 1.25 with and without the PAYGO pension system. For each simulation, given the parameter values of Table 1 (including β), the capital-output ratio in the initial steady state is allowed to differ from the baseline value and so also the initial real interest rate value.⁴⁵

As expected, compared to the baseline, the demographic impact on the real interest rate is more negative for values of ρ smaller than 1, less negative for values bigger than 1. The numbers underlying these series are reported in Table 3 as percentage points change between 1990 and 2030 (the table also reports the initial steady state value of the real interest rate, r_0 , in the different simulations). When ρ takes the mid-point value in the range suggested by the empirical literature, $\rho \in [0.4, 0.6]$, the demographic impact on the real interest rate between 1990 and 2030 is -1.34 (-1.65) percentage points for the case with (without) PAYGO pension system. Values very close to the ones found under the baseline scenarios. However the real interest rate stands systematically at lower levels. Across comparable scenarios, there is a gap between the simulated series of about 0.5

⁴⁴Cf. Rognlie (2014) for an assessment of Piketty (2014) in relation to ρ .

⁴⁵Notice that the steady state real interest rate is: $r = \psi(K/Y)^{-\rho} - \delta$, so that targeting the same capital-output ratio of the baseline would give a different value of the real interest rate when ρ is different from the baseline value.

percentage points that widens after 2030. Overall, despite higher substitutability between labor and capital dampens the baseline negative impact of the demographic transition on the real interest rate, the sign is not reversed for any sensible value of ρ .⁴⁶ In fact, for values of ρ in the range reported by the empirical literature the demographic impact on the real interest rate is even more negative than what estimated in the baseline specification above, particularly in the period between 1990 and 2030.

	ρ	r_0	$\Delta r_{1990-2030}$
	0.5	5.70	-1.34
PAYGO	1	6.19	-1.30
	1.25	6.55	-1.23
	0.5	5.66	-1.65
NO PAYGO	1	6.19	-1.56
	1.25	6.56	-1.43

Table 3: Different values of ρ : impact on the real interest rate, 1990-2030 percentage points change

4.4.3 Age-dependent labor efficiency

The exogenous aggregate labor supply, $L_t = \sum_j h_j N_{t,j}$, is composed by two elements: (a) the number of people in the labor force, in the baseline assumed to be those that are between 15 and 64 years old; (b) the age-dependent labor supply in units of efficiency, h_j , obtained by interacting the profiles for productivity and participation rate (see Figure A.3). The final profile h_j is such that each person on average across all the ages is able to provide one unit of labor input. For example, between age 30 and 50 individuals provide more than twice as much labor as in the remaining periods of life, because they are either more productive or more participative in the labor market providing more worked hours, or both. Therefore, as the demographic distribution varies over time also the aggregate efficiency of labor varies over time. To isolate the impact of this time-varying aggregate efficiency, Figure 8 plots the series for output growth and the real interest rate under the hypothetical case that for each person in the working age the labor efficiency is equal to the average, namely to 1, so that in this case the aggregate labor supply is simply captured by the total number of people in the working age: $L_t = \sum_{j=0}^{j_r-1} N_{t,j}$.

Looking at the output growth rates in Figure 8, it is apparent that when one does not adjust for the age-dependent labor efficiency the impact of aging is more negative in the periods before 2013, while it is attenuated afterwords. The reason is that before the turning point the demographic distribution is tilted towards relatively younger cohorts, more productive and participative according to the profile h_j . The opposite occurs afterwards when a greater mass of those in the labor force

⁴⁶Simulations for values of $\rho > 1.25$ up to value 2, not reported here, confirm that aging exerts a downward pressure on the real interest rate in all cases of comparable magnitude across the simulations, while the starting level can be significantly higher.



Figure 8: Transition dynamics in EA12: equal age-dependent efficiency units of labor h_j

Note. Calibration of Table 1 with $h_j = 1$ for all $j = 0, 1, ..., j_r$. The discount factor β is recalibrated to be consistent with the target capital-output ratio of 2.1 in the initial steady state. The resulting β has value 0.9489.

is relatively older. So the growth rate of the aggregate labor input is relatively smaller (higher) before (after) 2013 when it is not evaluated with the age-dependent labor efficiency. This translated directly into lower (higher) output growth.

When the individual labor supply is flat at the value of 1, there is less scope for savings over the life cycle (even though in this case they do not borrow in the first part of their life). The saving rate tends to be systematically lower than in the baseline. This is reflected in a systematically higher real interest rate. Furthermore, as the aggregate labor supply varies by less along the transition, the real interest rate also varies by less. Between 1990 and 2030 the real interest rate decreases by only -0.84 percentage points (compared to -1.3 in the baseline).⁴⁷

4.4.4 Type of pension system

The type of pension system might significantly matter for the results. In the baseline analysis above, it is assumed that the PAYGO pension system has a replacement rate fixed at 45% in all periods of the transition while the tax rate adjusts in order for the government to have the budget

⁴⁷Figure A.8 shows the demeaned series over the period 1960-2020 decomposing the impact of demographic change into "labor quantity", stemming from the simulation where the aggregate labor supply is simply given by the number of people in the working age, and a residual component that captures the evolving "labor efficiency", i.e. how the efficiency profile h_j interacts with the number of people in the economy making the aggregate labor input more or less scarce. It shows that, compared to the mean over 1960 to 2020, before the 2000s both "labor quantity" and "labor efficiency" exert an upward pressure on the real interest rate; afterwards they both turn negative. It is remarkable how the results from the OLG model align with the results from the econometric model by Fiorentini et al. (2018)

balanced in each period (see equation (2.10)). Figure 9 reproduces the baseline path of the real interest rate and shows the associated increase in the tax rate under such a pension scheme. The tax rate increases by about 14 percentage points in the long-run, going from about 6% in 1950 to more than 20% in 2080. The path of the tax rate along the transition follows closely the evolution of the old-dependency ratio, the number of retirees over the number of effective workers in the economy.

The dash-dotted lines in Figure 9 show what happens when the government runs a different pension system. Instead of fixing the replacement rate and varying the tax rate over time, the government fixes the tax rate at the value of 6.21% (the initial steady state value corresponding to a replacement rate of 45%) and allows the replacement rate to varies endogenously in such a way, again, that its budget is balanced in each period. It results that the replacement rate decreases significantly going to slightly less than 12% in the long-run. Also the evolution of the replacement rate is directly connected to the evolution of the old-dependency ratio. The resulting path of the real interest rate when the replacement rate adjusts endogenously is almost identical to the case in which there is no pension system in the economy.⁴⁸



Figure 9: Transition dynamics in EA12: different pension schemes

Note. Calibration of Table 1 with two different pension schemes: (a) in the baseline (continuous lines) the replacement rate ρ is fixed at 45% while the tax rate τ_t varies over time such that the government budget is balanced in each period (see equation (2.10)); (b) in the second scheme (dashed-dotted lines) the tax rate is fixed at the initial steady state value of 6.21% while the replacement rate is allowed to vary over time such that the government budget is balanced in each period (see equation (2.10)).

4.4.5 Retirement age

The relative scarcity of the labor supply in the economy is determined not only by the age-dependent efficiency of labor but also by the retirement age (here exogenously imposed) which in turn influ-

⁴⁸These results are in line with the literature, cf. e.g. section 5.5. of Krueger and Ludwig (2007).
ences the overall willingness to save and, in particular, the pervasiveness of the pension system effects. What happens when individuals are allowed to retire later or earlier compared to the base-line (age 65)? Figure 10 offers an answer.

When the retirement age is increased by 10 years, so that all individuals retire at age 75 – a big policy shift – general equilibrium forces lead the path of the real interest rate to be essentially the same as in the baseline. The same occurs when the retirement age is lowered to age 60. Given the parametrization of the efficiency age-profile h_j , the associated increase (decrease) of aggregate labor efficiency when the retirement age is increased (decreased) is not material enough to determine a significant modification of the consumption-savings decisions.⁴⁹ On the one hand, an increase of the retirement age makes labor less scarce making firms demanding more capital. This should put upward pressure on the real interest rate. On the other hand, an increase of the retirement age reduces the total amount of pension benefits to be financed via the PAYGO system so that the tax rate reduces, implying smaller crowding-out effects on private savings, thus a smaller real interest rate. These two effects seem to compensate each other (with the flipped side when the retirement age is decreased) so that the impact on the real of a policy shift on the retirement age is nil.

Output per capita growth tends to be higher (lower) when the retirement age is increased (decreased) in the acutest phase of the demographic transition. If in 2006 the output per capita growth is basically the same in the three scenarios (retirement age at 65, 75, 60) standing at about -0.15% on a annual basis, in 2020 the model predicts that demographic change leads to a growth rate of -0.55% in the baseline (retirement age at 65), -0.45% when the retirement age is at 75, -0.75% when it is at 60. A magnitude that, arguably, is insufficient to justify a huge policy shift such as an increase of the retirement age by 10 years.

The dashed lines show what happens when not only the retirement age is increased by 10 years but also individuals are made equally efficient in production. In this way the baby boomers not only retire 10 years later but they are not a drag on the economy when they approach retirement as they are assumed to be equally efficient in production of e.g. a 30 years old. As a consequence in 2020 the impact of demographic change on output per capita growth is about the same as the one in 2006. Nonetheless, the model predicts that no matter which scenario considered by 2040 demographic change leads the growth rate of output per capita at about -0.5%.

The real interest rate, on the contrary, stands persistently at a higher level when the individuals in the economy are equally efficient across the ages. More so the higher the retirement age. To strengthen the point, the thin lines in Figure 10 show the results from the extreme case in which individuals work continuously until they reach the death age and are equally efficient irrespective of age. In this case the growth rate of the aggregate labor supply is equal to the population growth rate. As discussed also in section 4.2, in this case there is really no shock induced by aging to the output

⁴⁹This is confirmed when one allows the efficiency profile not to be re-normalized such that the average efficiency along the life-cycle is 1 when the retirement age is varied. The thin dashed line in Figure 10 shows that the increase in average efficiency associated with individuals staying ten more years in the labor force (it goes from 1 to about 1.0752) does not lead to any meaningful variation in the simulation series.



Figure 10: Transition dynamics in EA12: different retirement age j_r

Note. Calibration of Table 1 with different retirement ages j_r . In each simulation the discount factor β is recalibrated to be consistent with the target capital-output ratio of 2.1 in the initial steady state.

per capita growth which mildly fluctuates around zero. The decrease of the real interest rate is immaterial, about -0.18 percentage points between 1990 and 2030, associated with the lengthening of the survival rates.

These results suggest that, to mitigate the negative impact of aging on both the growth rate of output per capita and the real interest rate, increasing the retirement age *per se* does not do a satisfactory job. The main reason resides in the hump-shaped nature of the productivity-participation profile across the ages so that increasing the number of people in the labor force at the margin does not sufficiently reduce the effective scarcity of labor in the economy. A policy that not only increased the retirement age but also bent upwards such a hump-shaped profile by making older individuals more productive would provide a more satisfactory outcome.

5 Comparing the model with the data: the role of TFP growth

In the attempt to compare the model with the data at least three orders of considerations arise.

First, to have non-zero long-run economic growth (and to match the historical levels of trend growth) one needs to introduce at least one source of growth other than demographics. Following a standard view, that one can find in most macroeconomic textbooks, in this section the approach is to allow for non-zero exogenous *total-factor-productivity* (TFP) growth.

Second, the natural level of both the real interest rate and output are unobservable variables. As

such, one needs to employ econometric estimates to compare the model's results with.

Third, the concept of natural interest rate revolves around a *single* measure, with no difference across asset classes. In this regard, the OLG model employed here seems appropriate in allowing savings in a single asset whose return in equilibrium, coherently with Wicksellian theory, does not differ from the marginal product of capital which features as the natural rate (given the absence of nominal variables and frictions in the model).⁵⁰ Hence, this paper embraces the approach by Rachel and Summers (2019b) who find empirically a common downward trend across different types of assets⁵¹ and thus assert: " [...] for analyzing long-term trend movements in neutral real rates, it is appropriate to focus on factors relating to saving and investment propensities rather than issues of liquidity or risk".

This section introduces a new calibration with non-zero TFP growth which allows to be consistent with recent econometric estimates of natural outcomes (real interest rate and output) for the euro area. The simulation results from this new calibration are compared not only with econometric estimates of natural outcomes but also with actual data pertaining to estimates of the marginal product of capital.⁵²

The comparison suggests a broad consistency between the OLG model's simulation results, econometric estimates and data from different sources, providing evidence in support of a prominent role of demographics to explain low-frequency movements of real interest rates and output. Nonetheless, a certain degree of arbitrariness seems thus far inevitable when it comes to measuring the marginal product of capital in the data so that its level is likely to remain debatable.⁵³ Arguably, the results from the model's simulation on the real interest rate are better interpreted in terms of *variation* over time associated with demographic change (and TFP growth).⁵⁴ Empirical measures of the marginal product of capital reported below shall be interpreted accordingly.

⁵⁰See Anderson (2005) on the interpretation of Wicksell's natural rate.

⁵¹Specifically, by means of a principal components analysis they find that in a set of various US real yields (spanning government debt, corporate bond and equity markets) "the first principal component, which picks up the downward trend visible in all returns, explains 94% of the total variance in the underlying series" (Rachel and Summers, 2019b). ⁵²Appendix G compares the OLG model's simulations also with measures of real yields for a set of different assets.

⁵³Gomme et al. (2011), Gomme et al. (2015), Caballero et al. (2017), Marx et al. (2018) all argue in favor of an increasing discrepancy between real interest rates and real returns to productive capital. On the contrary, most contributions in the OLG model literature do not take this potential discrepancy into consideration by targeting either a measure of the marginal product of capital (like in section 2, cf. e.g. Domeij and Flodén (2006), Krueger and Ludwig (2007), Attanasio et al. (2007), Ludwig et al. (2012), De la Croix et al. (2013)) or a comparatively lower measure coming from data on real government bonds yields (cf. e.g. Carvalho et al. (2016), Gagnon et al. (2016), Bielecki et al. (2018), Jones (2018), Aksoy et al. (2019)). Eggertsson et al. (2019) also enter in this second set of contributions but allow for markups that can drive a wedge between the marginal product of capital and the interest rate. Ludwig et al. (2016) and Sudo and Takizuka (2019) represent exceptions allowing households savings in more than one asset.

⁵⁴This view is reinforced in OLG models such as those of Ludwig et al. (2016) and Sudo and Takizuka (2019) where it is generally found that the impact of demographic change along the transition dynamics does not differ much across different types of assets, while the initial level of the associated returns can differ markedly. Sudo and Takizuka (2019) report: "While the returns on capital r_t^K and the return on government bonds r_t^B are sizably different in level, [...], the difference is negligible when comparing the effect of demographic changes on the two rates".

5.1 Experiment and calibration with non-zero TFP growth

To compare the model's results with the data one needs to introduce non-zero long-run growth in the model. To do so, the approach here adopted is to have non-zero exogenous long-run total-factor-productivity (TFP) growth. Consider the following Cobb-Douglas production function (a special case of the function in equation (2.7)):

$$Y_t = Z_t K_t^{\psi} L_t^{1-\psi}$$
(5.1)

where $Z_t = g_t^Z Z_{t-1}$ is the TFP level assumed to grow exogenously at gross rate g_t^Z . The rest of the model is the same one described in section 2.

The experiment is the same perfect-foresight exercise described in section 4.1. Here, however, in addition to demographics also TFP growth is exogenously time-varying. The transition dynamics occurs between an initial and a final stationary equilibrium characterized by constant non-zero TFP growth. Such an equilibrium is generally referred to as *balanced-growth path* (BGP) as variables grow at a constant rate over time.⁵⁵ In the simulations TFP growth is assumed either to be constant along the whole transition or to come from the (filtered) Solow residual obtained via a standard growth regression.⁵⁶

Table 4 summarizes the parameter values in the calibration that allows to compare the model with the data. The calibration is done in order to be consistent with the levels of the real interest rate and output that find correspondence in the econometric estimates of the "natural rate of interest" and the "trend growth rate of the natural rate of output" in Holston et al. (2017) (HLW, henceforth) for the euro area.⁵⁷ To this end, in the baseline the TFP growth rate is set to $g_t^Z = g^Z = (1.017)^{1-\psi}$ for all periods t, namely to the long-run growth rate of total output of 1.7% per annum.⁵⁸

The annualized estimates of the natural interest rate for the euro area by HLW range from a maximum of about 3.2% in 1973 to a minimum of -0.2% in 2014. To be consistent with this range of values in those years the initial level of the real interest rate is targeted to be about 3%. Furthermore, to be consistent with data the initial investment-output ratio is targeted to be 0.2 (equally to the calibration in section 4.1). The bias towards capital in the production function is set to the relatively low level of 0.22 to have an implied depreciation rate of capital which does not deviate much from what assumed in the literature (generally about 10% per annum, see section

⁵⁵On a BGP the variables $\{c_t, a_t, d_t, w_t, K_t, C_t, Y_t\}_{\forall t}$ grow at rate $(g^Z)^{1/(1-\psi)}$. See Appendix C.

⁵⁶Furthermore, it is assumed that TFP growth reverts in the long-run to its unconditional mean determined by the estimation of an autoregressive process of order 1, AR(1). See Appendix E.

⁵⁷The latest estimates are updated online, see https://www.newyorkfed.org/research/policy/rstar. The law of motion for the natural rate of interest in HLW is assumed to be: $r_t^* = g_t + z_t$, where g_t is the trend growth rate of the natural rate of output and z_t captures other determinants of r^* .

⁵⁸Across different simulations it is found that $g^Z = (1.017)^{1-\psi}$ allows to broadly match the dynamics of trend growth reported by HLW for the euro area in the period 1972–2018, with little discrepancies depending on the range of sensible values chosen for the other parameters in the model. Notice that to impose long-run growth $(g^Z)^{1/(1-\psi)} =$ 1.017 one needs to impose a TFP growth rate $g^Z = (1.017)^{1-\psi}$, $0 < \psi < 1$.

4.1).⁵⁹ It results $\delta = 0.11$.⁶⁰

The value of the individual discount factor consistent with these parameter values and the exogenous demographics in the initial BGP is found numerically (by means of a solver for non-linear systems) and results: $\beta = 0.994878$.

Parameter	Value	Note
Structural:		
ho	1	capital-labor substitutability, Cobb-Douglas production function
	0.22	bias towards capital in the production function
δ	0.11	depreciation rate of capital
σ	1	constant relative risk aversion (CRRA), log preferences
$ar{d}$	0.45	net replacement rate of PAYGO pension system
β	0.994878	individual discount factor, endogenous in the initial BGP: $\bar{r} = 0.0297, \bar{\iota} = 0.2$
Demographic:		
J	84	terminal life-age (99)
j_r	50	retirement age (65)
h_j	Figure A.3	individual life-cycle effective labor supply. Source: Domeij and Flodén (2006)
$N_{t,0}$	Figure A.2	incoming cohort size (age 15). Source: UN (2017)
$\pi_{t+j,j}$	Figure A.2	unconditional survival probabilities, retrieved from UN (2017) applying (2.1)
Technology:		
g_t^Z	Figure E.1	TFP growth, $(1.017)^{1-\psi} \forall t$ in baseline experiment

Table 4: Calibration with non-zero TFP growth: parameter values

Before solving for the transition dynamics the final stationary equilibrium is solved numerically for the endogenous real interest rate prevailing with the long-run demographics, imposing the parameter values in Table 4. Two main final stationary equilibria are studied: one where g_t^Z is simply set to its initial value of $(1.017)^{1-\psi}$ – the same value imposed throughout the whole transition; another one where the final value of g_t^Z is the unconditional mean obtained by estimating an autoregressive process of order 1 on the Solow residual (see Appendix E). In the former case the long-run real interest rate is about 1.63%, in the latter (with g^Z such that the long-run output growth is about 1.2%) about 1.18%.⁶¹

⁵⁹Bielecki et al. (2018) calibrate a similarly low value, 0.25, for the bias towards capital in production. They target the average real interest rate of 1.2% for the euro area in the period 1999-2008.

⁶⁰Denote with an upper bar the targeted values, $\bar{r} = 0.0297$, $\bar{\iota} = 0.2$. On a BGP the real interest rate reads: $\bar{r} = \psi/\kappa - \delta$, where κ denotes the capital-output ratio: $\kappa = \bar{\iota}/[\delta + (g^Z)^{1/(1-\psi)} - 1]$ with $\bar{\iota}$ denoting the targeted investment-output ratio. Given $\psi = 0.22$, $g^Z = 1.017$, from these two equations it results that the implied depreciation rate of capital is: $\delta = \{\bar{r} - (\psi/\bar{\iota})[(g^Z)^{1/(1-\psi)} - 1]\}/(\psi/\bar{\iota} - 1) = 0.11$.

⁶¹One can easily compute the value of TFP growth that would allow to restore a level of 3% for the real interest rate once the demographic transition is ended. One needs to solve numerically the final stationay equilibrium for g^Z imposing a value of the final real interest rate \bar{r} equal to 0.03. It is found that the value of g^Z consistent with a target \bar{r} tends to be such that the growth rate of output equals the targeted real interest rate. The output growth rate in the long-run is given by $(g^Z)^{1/(1-\psi)} = (\bar{r} + \delta)(\bar{\iota}/\psi) - \delta + 1$. Across different values for \bar{r} the implied investment-output ratio in the long-run $\bar{\iota}$ tends to be at a value very close to 0.22, which is the value assumed for ψ . Hence $(g^Z)^{1/(1-\psi)} \approx 1 + \bar{r}$. Which also means that absent technology growth the final real interest rate implied by demographics only would be close to zero given the current calibration.

5.2 Results: model vs data

Figure 13 compares data and model's simulations on total output growth and investment-output ratio under different scenarios for the exogenous TFP growth rate.

Output growth. When the TFP growth rate is the "actual" Solow residual growth rate (see Appendix E) the correspondence between the model's simulations and data from the Area Wide Model (AMW) dataset on output growth is quite close because of the very way the Solow residual is constructed.⁶² In this case it is assumed that in the long-run the TFP growth series reverts to its unconditional mean (estimated via an AR(1) process in the period 1999-2018) which implies a long-run output growth of about 1.2%. The simulated series is at a minimum around year 2030 with a value of about 0.5% growth per annum.

Assuming that the TFP growth is constant throughout the whole transition (at a value which implies a long-run output growth of 1.7%) allows to match quite closely the estimate of trend growth provided by HLW, with discrepancies between the two series in the range of about \pm 0.3 percentage points. Before the 1970s, however, it generates values that seem too low if compared with the actual data (shaded area). Also in this case the series is at a minimum around year 2030 when it has a value of about 0.9% growth per annum.

Using the low frequency component of the Solow residual (as implied by the Hodrick-Prescott filter, see Appendix E) allows to have more sensible values before the 1970s. It returns a bleaker picture in all years since the late 1990s, also reaching a minimum around year 2030 at a value of about 0.3% growth per annum.

These results show that once a certain level of exogenous technology growth is accounted for, demographic change seems to be a key determinant of the low-frequency movements of total output growth which find close correspondence with econometric estimates such as those provided by HLW. Thus, the results from the OLG model allow to qualify the determinants of those agnostic econometric estimates and provide a forecast relying on demographic projections. As explained in the previous sections, aging in the model exerts a persistent downward pressure on output growth due to the increasing scarcity of effective labor at least until 2030. Holding fixed the long-run TFP growth, between 1990 and 2030 the output growth decreases by about 1.45 percentage point. When one takes into account also the decline in TFP growth (measured by the filtered Solow residual) the figure is about 2.04 percentage points decline over the same period.

Investment-output ratio. The investment-output ratio is quite stable throughout the transition if compared with the range of values in the data. It is remarkable how the model with both the actual and the filtered TFP growth allows to match quite closely the main variations in the data. The model predicts that in the projected horizon the investment output ratio goes from a value of

⁶²Since using the "actual" TFP growth leads the model's simulation series on output growth to be very close to the actual series, the model in this case is not capturing the potential or natural output growth. Nonetheless, it is a useful exercise to see the overall direction of the real interest rate so captured, especially given that in this way the OLG model provides the closest results with the estimates of the natural rate by HLW (see Figure 11).

about 22% in 2016 to about 21% in 2040 with a decline that has no precedents in the past according to the model, no matter which simulation's specification is used.

The close correspondence between data and model for output growth and investment-output ratio lays the basis to the model's results on the real interest rate whose counterpart in the data is more debatable.



Figure 11: Model vs data: real interest rate

Note. The long-run growth imposed in the final stationary equilibrium is denoted by $g_{LR} = (g^Z)^{1/(1-\psi)}$. See Appendix E for the Solow residual used in the simulations: actual and Hodrick-Prescott (HP) filtered. Econometric estimate for the euro area (EA) corresponding to the "natural rate of interest" in Holston et al. (2017) (HLW). The bright (yellow) shaded area shows the 90-10 percentile range of EA12 countries on a measure of the marginal product of capital (MPK) computed using data from the Penn World Table (PWT) with time varying depreciation rates. Details in Appendix F. The thin black lines mark a band corresponding to $g_{LR} = 1.7\% \pm 0.6\%$ in the OLG model's simulations.

Real interest rate. Figure 11 compares the model's simulation results with the econometric estimate of the natural interest rate by HLW for the euro area. In addition, it shows the 90-10 percentile range fora measure of the marginal product of capital net of depreciation (MPK) derived from the Penn World Table (PWT).⁶³ These results deserve few comments.

⁶³Details on measures and series constructions are provided in Appendix F. Appendix G makes a further comparison of the model's results with data showing a set of measures of the real interest rate that span: (*i*) yields on short- and long-term government securities; (*ii*) a measure of the return of capital which in addition to government securities yields incorporate equity dividend yields and housing rental returns with data provided in the Jordá et al. (2019) (JST)

First, the impact of demographic change on the real interest rate is largely unaffected compared with the calibration in section 4.1. When TFP growth is fixed in each period of the transition (at a value implying a long-run output growth of 1.7%) the real interest rate decreases by about 1.1 percentage points between 1990 and 2030, going from a value of about 2.24% in 1990 to 1.13% in 2030. Over the same period the decrease is about 1.86 percentage points (2.6% to 0.74%) if the model is augmented with the actual TFP growth rate, 1.65 percentage points (2.22% to 0.57%) with the HP filtered TFP growth rate.

Second, there is quite a good correspondence between the OLG model's simulations and the econometric estimate of HLW. This correspondence deteriorates after 2008 suggesting that forces other than demographics and TFP might be at play in this period (perhaps due to factors linked to liquidity and risk not captured by the OLG model), determining the low-frequency movements of the real interest rate in the econometric estimate. But this discrepancy might be also due to the relative inaccuracy with which the HLW methodology leads to estimate the natural rate compared to trend growth (see Box 3 in Brand et al. (2018)) and which might magnify its downward trend (see Buncic (2019)).

Third, the model augmented with the actual TFP growth produces a simulated series which mimic very closely the econometric estimate by HLW. But even without the cumbersome utilization of the actual TFP growth series, the overall trend generated by the OLG model captures the dynamics implied by the econometric estimate.⁶⁴ This provides support for the prominent role of demographics in explaining the downward trend in the natural rate generally found by econometric estimates, but left unexplained due to the very structure of the econometrics employed. In particular, no matter the long-run value of technological growth, the OLG model predicts a downward impact of aging on the real interest rate at least until 2030.⁶⁵

Finally, in the early part of the simulations, before the 1970s, the high TFP growth rate (which allows to match the observed growth rate of output, see Figure 13) translates inevitably in a relatively high level of the real interest rate. This level is still within the 90-10 percentile range for the MPK and consistent with the levels of the real yields in the data. But it is probably on the high side (especially if compared with the econometric estimate at the world level by DGGT, see Appendix G). Overall, one shall consider with caution the simulation results in the early part of the sample also because of possible front-loading effects that might be linked to the assumption that the system is at the initial stationary equilibrium in 1950.⁶⁶ Still, it is remarkable that the model's simulation

[&]quot;return on everything" dataset. It further plots the econometric estimates of the natural real interest rate of Del Negro et al. (2019) (DGGT) for the world economy.

⁶⁴The Pearson's correlation coefficient between the two series is about 0.8 in the 1972-2018 period, when the model has the actual TFP growth. This coefficient is about 0.75 and 0.76 for the model with the HP filtered TFP growth and with fixed TFP growth, respectively.

⁶⁵When the model assumes a long-run TFP growth such that in the final stationary equilibrium output growth is 1.7% (1.2%, 1.1%) and the implied real interest rate is about 1.63% (1.18%, 1.05%), the real interest rate in 2030 is 1.13% (0.74%, 0.57%). That is, irrespective to the long-run technology growth assumed, the real interest rate in 2030 is lower with respect to its long-run value of about 0.5 percentage points or slightly less.

⁶⁶Also, consider that due to lack of data for all years before 1955 in the transition the TFP growth rate has the same

series capture the overall trend found in the data for the MPK, in the early part of the sample too.

5.3 Isolating the drivers

To isolate the quantitative contribution of the main exogenous drivers in the model, Figure 12 reports the results from three simulations:⁶⁷ (i) using the trend TFP growth (which implies a long-run output growth of 1.1%); (ii) fixing the TFP growth rate at the initial level throughout the whole transition (which implies a long-run output growth of 1.7%); (iii) in addition to (ii), fixing the survival probabilities at their first period level throughout the whole transition. To facilitate comparison with section 4.3 (compare Figure 12 with Figure 5) the initial reference year is 1987, which is when the baseline series takes approximately its mean value over the 1980s (normalized to zero in the figure). The series from the first two simulations correspond to the lightblue and black lines in Figure 11, respectively, reported as percentage point deviation from the reference year in Figure 12. Similarly to what discussed in section 4.3, the third exercise allows to isolate the contributions hinging upon changes in the mortality and fertility rates over time.



Figure 12: Real interest rate in EA12: drivers

Note. Calibration of Table 4. The results are obtained by running the simulations fixing the conditional survival probabilities at the their first period values and assuming that the economy has the TFP growth of the initial steady state throughout the whole transition (cf. black continuous line in Figure 11). The reference year is 1987 (when the baseline series takes approximately its mean value over the 1980s, normalized to zero).

Compared to the average in the 1980s, in the baseline – with time-varying TFP growth rate

⁽relatively high) value it has in 1955, see Appendix E.

⁶⁷Compared to section 4.3 the only difference here is the presence of non-zero TFP growth.

(as captured by the trend component of the Solow residual) – the real interest rate is about 1.8 percentage points lower in 2030 when the series reaches its trough. Of these, about 0.55 percentage points are due to a lower TFP growth (cf. the "TFP" shaded area in Figure 12); the remaining 1.25 percentage points are due to demographics, of which 0.75 can be attributed to changes in the survival probabilities and the residual 0.5 to changes in the growth rate of the number of newborns in the model (cf. the "Mortality" and "Fertility" shaded areas in Figure 12, respectively).

Equally to what found in section 4.3, fertility is responsible for the rebouncing of the real interest rate after 2030 while the permanently higher survival probabilities and lower TFP growth rate lead the real interest rate in the very long-run to be permanently lower compared to the mean in the 1980s. In 2100 the real interest rate is about 1.5 percentage points lower than the mean in the 1980s, of which 0.55 are due to TFP growth and the remainder to the survival probabilities.

These results confirm the orders of magnitude reported in Figure 5, adding the contribution of TFP growth. If one admits that demographics and TFP growth are the main drivers of the real interest rate (given the closeness of the econometric and OLG model's results, cf. Figure 11), one can argue that demographics account for about two-thirds of the projected decline of the real interest rate in the euro area until 2030 compared to the 1980s average (cf. Bielecki et al. (2020)).



Figure 13: Model vs data: total output growth and investment-output ratio, euro area

Note. The long-run growth imposed in the final stationary equilibrium is denoted by $g_{LR} = (g^Z)^{1/(1-\psi)}$. See Appendix E for the Solow residual used in the simulations: actual and Hodrick-Prescott (HP) filtered. Total output growth econometric estimate corresponding to "trend growth rate of the natural rate of output" in Holston et al. (2017) (HLW). Data from the "Area Wide Model" (AMW) dataset refer to the annualized growth rate of the YER series (Gross Domestic Product (GDP) at market prices, Chain linked volume, Reference year 1995") and to the ratio of IER (Gross Fixed Capital Formation, Chain linked volume, Reference year 1995) over YER. The shaded area denotes the 90–10 percentile range for the twelve EA12 countries, where the series used are from the Penn World Table (PWT) national accounts: $q_{-g}dp$ ("GDP at constant national 2011 prices"), q_{-1} ("Investment at constant national 2011 prices"). The thin black lines mark a band corresponding to $g_{LR} = 1.7\% \pm 0.6\%$.

6 Concluding remarks

By means of a large-scale overlapping generation (OLG) model, this paper finds that demographic change has a significant impact on the natural real interest rate for the euro area: a slight upward pressure in the 70s and 80s and a prolonged and marked downward pressure that extends at least until 2030 as the aging process unfolds (according to UN (2017) demographic projections). The model predicts in the baseline a decrease of the natural real interest rate of about 1.4 percentage point going to 2030 compared to the average in the 1980s. This estimate is never significantly dampened according to a set of sensitivity specifications which, if anything, suggest that it is a conservative number. The results indicate that the downward impact of aging could be mitigated not only by higher substitutability in production between labor and capital and higher intertemporal elasticity of substitution in consumption, but also by reforms aiming particularly at increasing the relative productivity of older cohorts and the participation rate. Increasing the retirement age per se has a very limited, if not null, mitigating effect.

Two drivers prevail explaining why aging has a downward total impact on the natural real interest rate: effective labor supply becomes scarcer and individuals increase their willingness to save in anticipation of higher survival probabilities. The latter driver is found to have relative more prominence in explaining the downward pressure of aging on the natural real interest rate especially in the very long-run. The presence of a pay-as-you-go pension system with defined benefits can in part offset this downward pressure by crowding-out productive capital but cannot significantly overturn it.

Importantly, the simulation results from the OLG model finds correspondence in econometric estimates of natural outcomes (real interest rate and output growth). Hence the results from the OLG model allow to qualify the determinants of agnostic econometric estimates and provide a forecast relying on demographic projections.

Finally, while the ongoing demographic transition seems to have definite time-lags, that allow to have a relatively clear picture of demographics in the future, its macroeconomic impact going forward might be altered by fundamental changes in the underlying structure of the economy. Such changes might comprise not only variations of parameters that the model in this paper allowed to control for (such as potential changes in the elasticity of substitution between labor and capital in production, intertemporal elasticity of substitution in consumption, productivity by age), but also technological changes that might alter the very nature of the production function (for example via the adoption of automation technologies) or processes of endogenous growth involving investment in human capital, as well as issues related to public debt sustainability and social tenability. All these potential changes lie outside the scope of the model adopted in this paper but offer interesting avenues for future research speculations.

References

- Acemoglu, D. and Restrepo, P. (2018). Demographics and automation. Working Paper 24421, National Bureau of Economic Research.
- Aksoy, Y., Basso, H. S., Smith, R. P., and Grasl, T. (2019). Demographic Structure and Macroeconomic Trends. *American Economic Journal: Macroeconomics*, 11(1):193–222.
- Anderson, R. G. (2005). Wicksell's natural rate. *Economic SYNOPSES*, (6). St. Luis Fed.
- Attanasio, O., Kitao, S., and Violante, G. L. (2007). Global demographic trends and social security reform. *Journal of Monetary Economics*, 54(1):144–198.
- Auerbach, A. J. and Kotlikoff, L. J. (1987). Dynamic fiscal policy. *Cambridge University Press*. Cambridge, UK.
- Barany, Z., Coeurdacier, N., and Guibaud, S. (2018). Capital flows in an aging world.
- Bielecki, M., Brzoza-Brzezina, M., and Kolasa, M. (2018). Demographics, monetary policy and the zero lower bound. *National Bank of Poland Working Paper No* 284.
- Bielecki, M., Brzoza-Brzezina, M., Kolasa, M., et al. (2020). Demographics and the natural interest rate in the euro area. Technical report. SGH KAE Working Paper Series Number: 2020/050.
- Börsch-Supan, A., Härtl, K., and Ludwig, A. (2014). Aging in Europe: Reforms, International Diversification, and Behavioral Reactions. *American Economic Review*, 104(5):224–229.
- Brand, C., Bielecki, M., and Penalver, A. (2018). The natural rate of interest: estimates, drivers, and challenges to monetary policy. Occasional Paper Series 217, European Central Bank.
- Buncic, D. (2019). Econometric issues with laubach and williams' estimates of the natural rate of interest. manuscript.
- Caballero, R. J., Farhi, E., and Gourinchas, P.-O. (2017). Rents, Technical Change, and Risk Premia Accounting for Secular Trends in Interest Rates, Returns on Capital, Earning Yields, and Factor Shares. *American Economic Review*, 107(5):614–620.
- Carvalho, C., Ferrero, A., and Nechio, F. (2016). Demographics and real interest rates: Inspecting the mechanism. *European Economic Review*, 88:208–226.
- Caselli, F. (2005). Accounting for Cross-Country Income Differences. In Aghion, P. and Durlauf, S., editors, *Handbook of Economic Growth*, volume 1 of *Handbook of Economic Growth*, chapter 9, pages 679–741. Elsevier.
- Caselli, F. and Feyrer, J. (2007). The Marginal Product of Capital. *The Quarterly Journal of Economics*, 122(2):535–568.
- Chirinko, R. S. (2008). σ : The long and short of it. *Journal of Macroeconomics*, 30(2):671–686. June.

- Cooley, T. and Henriksen, E. (2018). The demographic deficit. *Journal of Monetary Economics*, 93(C):45–62.
- Cooley, T. F., Henriksen, E., and Nusbaum, C. (2019). Demographic Obstacles to European Growth. NBER Working Papers 26503, National Bureau of Economic Research, Inc.
- De la Croix, D., Pierrard, O., and Sneessens, H. R. (2013). Aging and pensions in general equilibrium: Labor market imperfections matter. *Journal of Economic Dynamics and Control*, 37(1):104–124.
- Del Negro, M., Giannone, D., Giannoni, M. P., and Tambalotti, A. (2019). Global trends in interest rates. *Journal of International Economics*, 118(C):248–262.
- Domeij, D. and Flodén, M. (2006). Population aging and international capital flows. *International Economic Review*, 47(3).
- Draghi, M. (2016). Addressing the causes of low interest rates. Intriductive speech delivered at the Annual Meeting of the Asian Development Bank, May 2.
- Eggertsson, G. B., Mehrotra, N. R., and Robbins, J. A. (2019). A Model of Secular Stagnation: Theory and Quantitative Evaluation. *American Economic Journal: Macroeconomics*, 11(1):1–48.
- Eichenbaum, M. S., Hansen, L. P., and Singleton, K. J. (1988). A Time Series Analysis of Representative Agent Models of Consumption and Leisure Choice Under Uncertainty. *The Quarterly Journal of Economics*, 103(1):51–78.
- Fagan, G., Henry, J., and Mestre, R. (2001). An area-wide model (AWM) for the euro area. Working Paper Series 42, European Central Bank.
- Feenstra, R. C., Inklaar, R., and Timmer, M. P. (2015). The Next Generation of the Penn World Table. *American Economic Review*, 105(10):3150–3182.
- Fernndez-Villaverde, J. and Krueger, D. (2007). Consumption over the Life Cycle: Facts from Consumer Expenditure Survey Data. *The Review of Economics and Statistics*, 89(3):552–565.
- Fiorentini, G., Galesi, A., Pérez-Quirós, G., and Sentana, E. (2018). The rise and fall of the natural interest rate. *Banco de España*. Draft Paper.
- Fullerton, H. N. (1999). Labor force participation: 75 years of change, 1950-98 and 1998-2025. *Monthly Labor Review*, 122:3–12.
- Gagnon, E., Johannsen, B., and Lopez-Salido, J. (2016). Understanding the new normal: The role of demographics. *FEDS Working Paper*.
- Gagnon, E., Johannsen, B., and Lopez-Salido, J. (2018). Comment on the demographic deficit. *Journal of Monetary Economics*, 93(C):63–67.
- Gertler, M. (1999). Government debt and social security in a life-cycle economy. *Carnegie-Rochester Conference Series on Public Policy*, 50(1):61–110.

- Gomes, S., Jacquinot, P., and Pisani, M. (2012). The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area. *Economic Modelling*, 29(5):1686–1714.
- Gomme, P., Ravikumar, B., and Rupert, P. (2011). The Return to Capital and the Business Cycle. *Review of Economic Dynamics*, 14(2):262–278.
- Gomme, P., Ravikumar, B., and Rupert, P. (2015). Secular Stagnation and Returns on Capital. *Economic Synopses*, (19).
- Goodhart, C. and Pradhan, M. (2017). Demographics will reverse three multi-decade global trends. *Bis Working Papers*, (656). cf. "Demography and Economics: Look Past the Past", VoxEU.org.
- Hansen, G. and Imrohoroglu, S. (2008). Consumption over the Life Cycle: The Role of Annuities. *Review of Economic Dynamics*, 11(3):566–583.
- Hansen, G. D. (1993). The cyclical and secular behaviour of the labor input: Comparing efficiency units and hours worked. *Journal of Applied Econometrics*, 8:71–80.
- Havranek, T. (2013). Publication Bias in Measuring Intertemporal Substitution. Working Papers IES 2013/15, Charles University Prague, Faculty of Social Sciences, Institute of Economic Studies.
- Holston, K., Laubach, T., and Williams, J. C. (2017). Measuring the natural rate of interest: International trends and determinants. *Journal of International Economics*, 108(S1):59–75.
- Jones, C. (2018). Aging, Secular Stagnation and the Business Cycle. IMF Working Papers 18/67, International Monetary Fund.
- Jordá, O., Knoll, K., Kuvshinov, D., Schularick, M., and Taylor, A. M. (2019). The Rate of Return on Everything, 18702015. *Quarterly Journal of Economics*, 134(3):1225–1298.
- Kara, E. and von Thadden, L. (2016). Interest rate effects of demographic changes in a newkeynesian life-cycle framework. *Macroeconomic Dynamics*, 20(1):120–164.
- Karabarbounis, L. and Neiman, B. (2014). The Global Decline of the Labor Share. *The Quarterly Journal of Economics*, 129(1):61–103.
- Keynes, J. M. (1937). Some economic consequences of a declining population. *The Eugenics Review*, 29(1):13–17.
- Krueger, D. and Ludwig, A. (2007). On the consequences of demographic change for rates of returns to capital, and the distribution of wealth and welfare. *Journal of Monetary Economics*, 54(1):49–87.
- Lee, R. (2016). *Macroeconomics, Aging, and Growth*, volume 1B of *Handbook of the Economics of Population Aging*, chapter 7, pages 59–118. Elsevier.
- Lee, R. and Mason, A., editors (2011). *Population Aging and the Generational Economy*. Number 13816 in Books. Edward Elgar Publishing.

- Lisack, N., Sajedi, R., and Thwaites, G. (2017). Demographic trends and the real interest rate. Bank of England working papers 701, Bank of England.
- Ludwig, A., Geppert, C., and Abiry, R. (2016). Secular Stagnation? Growth, Asset Returns and Welfare in the Next Decades. Annual Conference 2016 (Augsburg): Demographic Change 145764, Verein fr Socialpolitik / German Economic Association.
- Ludwig, A., Schelkle, T., and Vogel, E. (2012). Demographic Change, Human Capital and Welfare. *Review of Economic Dynamics*, 15(1):94–107.
- Marx, M., Mojon, B., and Velde, F. R. (2018). Why Have Interest Rates Fallen Far Below the Return on Capital. Working Paper Series WP-2018-1, Federal Reserve Bank of Chicago.
- Piketty, T. (2014). *Capital in the Twenty-First Century*. Cambridge Massachusetts: Belknap Press. March.
- Quadrini, V. and Ríos-Rull, J.-V. (2015). Chapter 14 inequality in macroeconomics. In Atkinson, A. B. and Bourguignon, F., editors, *Handbook of Income Distribution*, volume 2 of *Handbook of Income Distribution*, pages 1229 – 1302. Elsevier.
- Rachel, L. and Summers, L. H. (2019a). On Falling Neutral Real Rates, Fiscal Policy, and the Risk of Secular Stagnation. *Brookings Papers on Economic Activity*.
- Rachel, L. and Summers, L. H. (2019b). On Secular Stagnation in the Industrialized World. NBER Working Papers 26198, National Bureau of Economic Research, Inc.
- Rognlie, M. (2014). A note on piketty and diminishing returns to capital. *mimeo*. http://mattrognlie.com/piketty_diminishing_returns.pdf.
- Solow, R. M. (1956). A Contribution to the Theory of Economic Growth. *The Quarterly Journal* of Economics, 70(1):65–94.
- Sudo, N. and Takizuka, Y. (2019). Population ageing and the real interest rate in the last and next 50 years – a tale told by an overlapping generations model –. *Macroeconomic Dynamics*. Forthcoming. Previous veersion: Bank of Japan Working Paper Series No. 18-E-1, January.
- UN (2017). World Population Prospects: The 2017 Revision. United nations, department of economics and social affairs, population division, The 2017 Revision, DVD Edition.
- Wicksell, K. (1898). *Interest and Prices*. Nihon Keizai Hyoron Sha. Translated by R.F. Kahn as Interest and Prices (1936), orignal title: Geldzins und Guterpreise.
- Woodford, M. (2003). *Interest and Prices: Foundations of a Theory of Monetary Policy*. Princeton University Press. Princeton, N.J.; Woodstock, Oxfordshire England.

Appendix

A Additional figures



Figure A.1: Old dependency ratio in Europe

Note. The indicator in the figure is the the number of people aged more than 64 over the number of people aged between 15 and 64. Data from the United Nations (UN, 2017) *World Population Prospects: The 2017 Revision*, medium variant after year 2016 (cf. footnote 18). The following groups of countries hold. EA19: Austria, Belgium, Cyprus, Estonia, Finland, France, Germany, Greece, Ireland, Italy, Latvia, Lithuania, Luxembourg, Malta, Netherland, Portugal, Slovakia, Slovenia, Spain; EA12 is EA19 excluding Cyprus, Estonia, Latvia, Lithuania, Malta, Slovakia, Slovenia; EA5: France, Germany, Italy, Netherlands, Spain; EU28 comprises EA19 and the following non-EA members: Bulgaria, Croatia, Czech Republic, Hungary, Poland, Romania, Sweden, Denmark and United Kingdom.



Figure A.2: Euro area (EA12) demographics in the model

Note. Data are from UN (2017) including medium variant projections until year 2100. The unconditional survival probabilities $\pi_{t+j,j}$ are retrieved applying the recursive formula $N_{t+j,j} = \pi_{t+j,j}N_{t,0}$ using data for the cohort size $N_{t,j}$ for each year t and age-bin j, with $N_{t,0}$ corresponding to the incoming cohort size, those aged 15. The marked lines correspond to $\pi_{\tau+j,j}$ for $\tau \in \{1951, 1980, 2010, 2040\}$ for all ages j in the model, the other lines capture the profiles in the remaining years.



Figure A.3: Age dependent labor supply in efficiency units, h_i

Note. The profile is obtained with a cubic interpolation (for age 15 to 70) on the data points provided in Domeij and Flodén (2006). These data points are the product of participation rates provided by Fullerton (1999) and productivity provided by Hansen (1993). Lacking data, for $j \ge 70$ the profile is obtained from the following logistic function: $C/(1 + Ae^{-Bj})$, with A = .49, C = 50, $B = (1/70) \log [h_{70}A/(C - h_{70})]$. The blue continuous line denotes the baseline profile with exogenous retirement age at jr + 1 = 65. For each exogenous retirement age, the productivity profile h_j is normalized across the ages such that its mean is equal to 1.



Figure A.4: Age profiles: initial versus final steady state

Note. Baseline calibration, see Table 1. The dotted lines show the hypothetical (partial equilibrium) final steady state scenario in which the real interest rate is fixed at the initial steady state value.



Figure A.5: Growth rate of output: different sample periods

Note. The sample means in the first panel refer to the baseline simulation: $\sigma = 1$ with PAYGO pension system.



Figure A.6: EA12 transition dynamics: main aggregates per capita

Note. Baseline calibration, see Table 1



Figure A.7: Individual consumption profiles: selected cohorts versus initial and final steady state

Note. Baseline calibration, see Table 1.



Figure A.8: Natural real interest rate, demeaned: theoretical vs econometric model: drivers

Note to A.8a. The real interest rate path of the model is obtained with the baseline specification of Figure 4. Since labor in efficiency units (L_t) depends on two parameters, their different impact is considered in isolation: "labor quantity" denotes the impact of the mere number of people in the labor force (aged between 15 and 64); "labor efficiency" denotes the impact of the age-varying productivity (technically, it is obtained by running the model twice: first, when L_t is the actual one: $\sum_{j=0}^{jr} h_j N_{t,j}$; second, considering $L_t = \sum_{j=0}^{jr} N_{t,j}$, i.e. by setting $h_j = 1$ for all j (see section 4.4.3). Finally, the difference between the two implied curves is taken). All the implied series are demeaned over the period 1960–2020 in order to produce the standardized results in the figure.

Note to A.8b. Estimates of the natural real interest rate provided by Fiorentini et al. (2018) for the Brand et al. (2018)'s report from a panel error correction model (ECM) at annual frequency over the period 1899-2016. The unbalanced panel of advanced economies includes the following 17 countries: Australia, Belgium, Canada, Denmark, Finland, France, Germany, Italy, Japan, Netherlands, Norway, Portugal, Spain, Sweden, Switzerland, United Kingdom, and US. The observed real rate serves as dependent variable in the ECM, while some indicators about total factor productivity, demographics and risk serve as regressors. Data on TFP growth comes from Penn World Tables and Total Economy Database by The Conference Board https://www.conference-board.org/data/economydatabase/; data on demographic composition come from the Human-Mortality-Database http://mortality.org. The spread between long-term and short-term interest rate is used as proxy for the term premium to measure the time-varying risk aversion of agents. Interest rates data come from the Jordá et al. (2019) Macro history Database and from the OECD Main Economic Indicators database.

B OLG model with zero technology growth

B.1 Set of equilibrium equations

In an equilibrium with perfect for esight, the following optimal conditions hold (labor augmenting technology $A_t = 1 \ \forall t$).

Households:

for
$$j = 0, 1, ..., J - 1$$
: $c_{t+1,j+1} = [\beta(1 + r_{t+1})]^{\frac{1}{\sigma}} c_{t,j}$ (B.1)

for
$$j = 0, 1, ..., J$$
: $c_{t,j} + a_{t+1,j+1} = \frac{a_{t,j}(1+r_t)}{s_{t,j}} + w_t h_j (1-\tau_t) I(j < j_r) + d_t I(j \ge j_r)$ (B.2)
with: $a_{t,0} = 0$
 $a_{t,J+1} = 0$

Government:

$$\tau_t = \frac{d\overline{h} \sum_{j=j_r}^J N_{t,j}}{L_t + d\overline{h} \sum_{j=j_r}^J N_{t,j}}$$
(B.3)

$$d_t = d\bar{h}w_t(1-\tau_t)$$
(B.4)

Firms:

$$r_t + \delta = \left(\frac{Y_t}{K_t}\right)^{\frac{1}{\rho}}$$
 (B.5)

$$w_t = (1 - \psi) \left(\frac{Y_t}{L_t}\right)^{\frac{1}{\rho}}$$
(B.6)

Clearing:⁶⁸

$$K_{t+1} = A_t, \quad A_t = \sum_{j=0}^J a_{t+1,j+1} N_{t,j}$$
 (B.7)

$$C_t + K_{t+1} - (1 - \delta)K_t = Y_t, \quad C_t = \sum_{j=0}^J c_{t,j} N_{t,j}$$
 (B.8)

Check: 177 equations for 177 endogenous unknowns (J = 84):

$$\{c_t\}_{j=0}^J, \{a_t\}_{j=1}^J, r_t, w_t, \tau_t, d_t, K_t, \mathcal{A}_t, C_t, Y_t$$

⁶⁸Notice that the current decision on savings determine the next-period level of capital. That is, as it is standard, capital is a predetermined variable.

B.2 Steady state

A steady state is defined as a stationary equilibrium where the exogenous demographic variables assume a fixed value forever. Economic growth is zero in such equilibrium with zero technology growth. Variables in steady state have no time subscript. The following system of optimal conditions hold.

$$\begin{split} L &= \sum_{j=0}^{J} h_{j} N_{j} \\ r &= \left(\frac{K}{Y}\right)^{-\rho} - \delta \\ w &= (1-\psi) \left(\frac{r+\delta}{J}\right) \left(\frac{K}{L}\right)^{\frac{1}{\rho}} \\ \tau &= \frac{d\bar{h} \sum_{j=j_{r}}^{J} N_{j}}{L+d\bar{h} \sum_{j=j_{r}}^{J} N_{j}}, \quad \bar{h} = \sum_{j=0}^{J} h_{j} / j_{r} \\ d &= d\bar{h} (1-\tau) w \\ y_{j} &= (1-\tau) w h_{j} I (j < j_{r}) + dI (j \ge j_{r}), \quad \text{for } j = 0, 1, ..., J \\ c_{0} &= \frac{\sum_{j=0}^{J} \left(\frac{1}{1+r}\right)^{j} \pi_{j} (\beta(1+r))^{\frac{j}{\sigma}}}{\sum_{j=0}^{J} \left(\frac{1}{1+r}\right)^{j} \pi_{j} (\beta(1+r))^{\frac{j}{\sigma}}} \\ c_{j+1} &= [\beta(1+r)]^{\frac{1}{\sigma}} c_{j}, \quad \text{for } j = 0, 1, ..., J - 1 \\ a_{j+1} &= \frac{a_{j}(1+r)}{s_{j}} + y_{j} - c_{j}, \quad \text{for } j = 0, 1, ..., J \\ a_{0} &= 0 \\ a_{J+1} &= 0 \\ C &= \sum_{j=0}^{J} c_{j} N_{j} \\ K &= \sum_{j=0}^{J} a_{j} N_{j} \\ Y &= \left[\psi K^{\frac{\rho-1}{\rho}} + (1-\psi) L^{\frac{\rho-1}{\rho}}\right]^{\frac{\rho}{p-1}} \\ C + \delta K &= Y \end{split}$$

Exogenous demographics: $\{N_j\}_{j=0}^J$, from which the survival probabilities are derived: $s_j = N_j/N_{j-1}$ with $N_{-1} = N_0$ (such that $s_0 = 1$), $\pi_j = \prod_{k=0}^j s_k$; parameters: $\{h_j\}_{j=0}^J$, j_r , ψ , δ , \bar{d} , β , σ , ρ .

C OLG model with non-zero TFP growth

C.1 Set of scaled equilibrium equations

Consider the model with production function (5.1): $Y_t = Z_t K_t^{\psi} L_t^{1-\psi}$, $Z_t = g_t^Z Z_{t-1}$. To have stationarity in presence of technology (TFP) growth g_t^Z , each variable $x_t \in \{c_t, a_t, d_t, K_t, C_t, Y_t\}$ is scaled by the level of technology and effective labor:⁶⁹

$$\widetilde{x}_t = x_t / (Z_t^{\frac{1}{1-\psi}} L_t)$$

Further, have $g_t^L = L_t/L_{t-1}$ and $\widetilde{w}_t = w_t/(Z_t^{\frac{1}{1-\psi}})$. Then the following optimal conditions hold:

Households:

for
$$j = 0, 1, ..., J - 1$$
: $\widetilde{c}_{t+1,j+1} = \left[\beta(1+r_{t+1})\right]^{\frac{1}{\sigma}} \widetilde{c}_{t,j} \left[(g_{t+1}^Z)^{\frac{1}{1-\psi}} g_{t+1}^L \right]^{-1}$ (C.1)

for
$$j = 0, 1, ..., J$$
: $\widetilde{c}_{t,j} + \widetilde{a}_{t+1,j+1} (g_{t+1}^Z)^{\frac{1}{1-\psi}} g_{t+1}^L = \widetilde{a}_{t,j} (1+r_t)/s_{t,j} + \widetilde{w}_t (h_j/L_t) (1-\tau_t) I(j < j_r) + \widetilde{d}_t I(j > j_r)$ (C.2)

with:
$$\tilde{a}_{t,0} = 0, \quad \tilde{a}_{t,J+1} = 0$$
 (C.3)

Government:

$$\tau_t = \frac{\overline{dh} \sum_{j=j_r}^J N_{t,j}}{L_t + \overline{dh} \sum_{j=j_r}^J N_{t,j}}$$
(C.4)

$$\widetilde{d}_t = \overline{d}(\overline{h}/L_t)\widetilde{w}_t(1-\tau_t)$$
(C.5)

Firms:

$$r_t + \delta = \frac{\widetilde{Y}_t}{\widetilde{K}_t} \tag{C.6}$$

$$\widetilde{w}_t = (1 - \psi)\widetilde{Y}_t \tag{C.7}$$

Clearing (notice: $\widetilde{\mathcal{A}}_t = \mathcal{A}_t / (Z_{t+1}^{\frac{1}{1-\psi}} L_{t+1}))$:

$$\widetilde{K}_{t+1} = \widetilde{\mathcal{A}}_t, \quad \widetilde{\mathcal{A}}_t = \sum_{j=0}^J \widetilde{a}_{t+1,j+1} N_{t,j}$$
 (C.8)

$$\widetilde{C}_{t} + \widetilde{K}_{t+1} (g_{t+1}^{Z})^{\frac{1}{1-\psi}} g_{t+1}^{L} - (1-\delta) \widetilde{K}_{t} = \widetilde{Y}_{t}, \quad \widetilde{C}_{t} = \sum_{j=0}^{J} \widetilde{c}_{t,j} N_{t,j}$$
(C.9)

⁶⁹A balanced-growth-path (BGP) is characterized by constant effective capital. To characterize what is effective capital, and thus find the scaling factor in the model, consider that on a BGP the capital-output ratio needs to be constant. Its inverse reads:

$$\frac{Y_t}{K_t} = \frac{Z_t K_t^{\psi} L_t^{1-\psi}}{K_t} = \left(\frac{K_t}{Z_t^{\frac{1}{1-\psi}} L_t}\right)^{\psi-1}$$

which is constant if and only if $\widetilde{K}_t \equiv K_t / (Z_t^{\frac{1}{1-\psi}} L_t)$ is constant.

C.2 Balanced growth path

A balanced-growth-path (BGP) is defined as a stationary equilibrium where the exogenous demographic variables and the exogenous growth rate of technology assume a fixed value forever. Scaled variables are constant over time so that each variable grow at a constant rate over time determined by the exogenous growth rate of technology. That is, for each $x_t \in \{c_t, a_t, d_t, \{y_{t,j}\}_{j=0}^J, K_t, C_t, Y_t\}$ their respective constant growth rate on a BGP is given by:⁷⁰

$$g_t^x = x_t / x_{t-1} = (g^Z)^{\frac{1}{1-\psi}} \quad \forall t$$

Variables on a BGP have no time subscript. The following system of optimal scaled conditions hold.

$$\begin{split} L &= \sum_{j=0}^{J} h_{j} N_{j} \\ \widetilde{w} &= (1-\psi) \left(\frac{1}{r+\delta}\right)^{1-\psi} \\ \tau &= \frac{d\overline{h} \sum_{j=j_{r}}^{J} N_{j}}{L+d\overline{h} \sum_{j=j_{r}}^{J} N_{j}}, \quad \overline{h} = \sum_{j=0}^{J} h_{j} / j_{r} \\ \widetilde{d} &= d(\overline{h}/L)(1-\tau) \widetilde{w} \\ \widetilde{y}_{j} &= (1-\tau) \widetilde{w}(h_{j}/L) I(j < j_{r}) + \widetilde{d}I(j \ge j_{r}), \quad \text{for } j = 0, 1, ..., J \\ \widetilde{c}_{0} &= \frac{\sum_{j=0}^{J} (\frac{1}{1+r})^{j} \pi_{j} \widetilde{y}_{j} (g^{Z})^{\frac{1}{1-\psi}}}{\sum_{j=0}^{J} (\frac{1}{1+r})^{j} \pi_{j} (\beta(1+r)))^{\frac{j}{\sigma}}} \\ \widetilde{c}_{j+1} &= [\beta(1+r)]^{\frac{1}{\sigma}} \widetilde{c}_{j} (g^{Z})^{-\frac{1}{1-\psi}}, \quad \text{for } j = 0, 1, ..., J - 1 \\ \widetilde{a}_{j+1} &= \{\widetilde{a}_{j}(1+r)/s_{j} + \widetilde{y}_{j} - \widetilde{c}_{j}\}(g^{Z})^{-\frac{1}{1-\psi}}, \quad \text{for } j = 0, 1, ..., J \\ \widetilde{a}_{0} &= 0 \\ \widetilde{a}_{J+1} &= 0 \\ \widetilde{C} &= \sum_{j=0}^{J} \widetilde{c}_{j} N_{j} \\ \widetilde{K} &= \sum_{j=0}^{J} \widetilde{a}_{j} N_{j} \\ \widetilde{K} &= \sum_{j=0}^{J} \widetilde{a}_{j} N_{j} \\ \widetilde{Y} &= \widetilde{K} \\ \widetilde{C} + [(g^{Z})^{\frac{1}{1-\psi}} - 1+\delta] \widetilde{K} &= \widetilde{Y} \end{split}$$

Exogenous demographics $\{N_j\}_{j=0}^J$, from which the survival probabilities are derived: $s_j = N_j/N_{j-1}$ with $N_{-1} = N_0$ (such that $s_0 = 1$), $\pi_j = \prod_{k=0}^j s_k$; parameters: $\{h_j\}_{j=0}^J, j_r, \psi, \delta, \bar{d}, \beta, \sigma, g^Z$.

⁷⁰Notice that the steady state studied in (B.2) represents simply a special case of BGP where the TFP growth rate is zero, i.e. $g^Z = 1$.

D Two-period version of the model

The two-period economy described in section 3 of the main text is fully characterized by the following equations.

Firms' first order conditions:

$$r_t + \delta = \psi k_t^{\psi - 1} \tag{D.1}$$

$$w_t = (1 - \psi)A_t k_t \tag{D.2}$$

Social security:

$$d_{t+1} = \tau_{t+1} w_{t+1} \frac{g_{t+1}^N}{s_{t+1,1}}$$
(D.3)

$$d_t = \bar{d}w_t(1-\tau_t) \tag{D.4}$$

Households' Euler equation and budget constraint:

$$c_{t+1,1} = \beta(1+r_{t+1})c_{t,0} \tag{D.5}$$

$$c_{t,0} + a_{t+1,1} = (1 - \tau_t)w_t$$
 (D.6)

$$c_{t+1,1} = \frac{a_{t+1,1}(1+r_{t+1})}{s_{t+1,1}} + d_{t+1}$$
 (D.7)

Capital market clearing and law-of-motion of capital:

$$g_{t+1}^{N}A_{t+1}k_{t+1} = a_{t+1,1}$$
(D.8)

$$g^{A}g^{N}_{t+1}k_{t+1} = (1-\delta)k_{t} + i_{t}$$
(D.9)

Analysis. To derive an expression for the households' savings supply, plug (D.6) into (D.7) and then use (D.5) to have:

$$\beta(1+r_{t+1})c_{t,0} = \frac{[(1-\tau_t)w_t - c_{t,0}](1+r_{t+1})}{s_{t+1,1}} + d_{t+1}$$

Using (D.3) and rearranging, it results:

$$c_{t,0} = \frac{1}{1+\beta s_{t+1,1}} \left[(1-\tau_t) w_t + \tau_{t+1} w_{t+1} \frac{g_{t+1}^N}{1+r_{t+1}} \right]$$

Equation (D.3) needs to be used also into (D.4) to have the equilibrium tax rate:

$$\tau_{t+1} = \frac{\bar{d}}{\bar{d} + \frac{g_{t+1}^N}{s_{t+1,1}}} \tag{D.10}$$

Hence, the household's savings from (D.6) read:

$$a_{t+1,1} = (1 - \tau_t) w_t \frac{\beta s_{t+1,1}}{1 + \beta s_{t+1,1}} - \tau_{t+1} w_{t+1} \frac{g_{t+1}^N}{(1 + r_{t+1})(1 + \beta s_{t+1,1})}$$

First use (D.2):

$$a_{t+1,1} = (1-\tau_t)(1-\psi)A_tk_t \frac{\beta s_{t+1,1}}{1+\beta s_{t+1,1}} - \tau_{t+1}(1-\psi)A_{t+1}k_{t+1}^{\psi}\frac{g_{t+1}^N}{(1+r_{t+1})(1+\beta s_{t+1,1})}$$

then use (D.8) and simplify to have:

$$k_{t+1} = (1 - \tau_t)(1 - \psi)k_t \frac{\beta s_{t+1,1}}{g^A g_{t+1}^N (1 + \beta s_{t+1,1})} - \frac{\tau_{t+1}(1 - \psi)k_{t+1}^{\psi}}{(1 + r_{t+1})(1 + \beta s_{t+1,1})}$$
(D.11)

Only two equations have not been used to arrive to this result: (D.1) which identifies the capital demand by firms and the law-of-motion of capital (D.9). The goal now is to characterize the economy in terms of capital demand by firms and capital supply by households where the unique endogenous variables are the real interest rate r_t and the aggregate saving rate. The latter equals the investment-output ratio (due to the assumption of closed economy) which from (D.9) reads:

$$\iota_t \equiv \frac{I_t}{Y_t} = \frac{i_t}{y_t} = \frac{g^A g_{t+1}^N k_{t+1} - (1-\delta)k_t}{k_t}$$
(D.12)

Consider a *balanced-growth path* (BGP) characterized by a constant effective capital stock: $k_t = k$ for all t. From (D.12) the saving rate on a BGP reads:

$$\iota = [g^{A}g^{N} - (1 - \delta)]k^{1 - \psi}$$
 (D.13)

where $k^{1-\psi}$ is the capital-output ratio. Plug (D.13) into (D.1) to have:

$$\iota^{D} : r = \frac{g^{A}g^{N} - (1 - \delta)}{\iota} - \delta$$
(D.14)

which represents the *firms' capital demand* ι^D on the ι, r -Cartesian plane.

To have a comparable expression on the households' side, add and subtract $(1 - \delta)k_t$ to (D.11) and evaluate on a BGP to have:

$$\frac{g^A g^N k - (1 - \delta)k}{k^{\psi}} = (1 - \tau)(1 - \psi) \frac{\beta s_1}{(1 + \beta s_1)} - \frac{g^A g^N \tau (1 - \psi)}{(1 + r)(1 + \beta s_1)} - (1 - \delta) \frac{k}{k^{\psi}}$$

Using (D.13), it reduces to:

$$\iota = \left(1 - \frac{1 - \delta}{g^A g^N}\right) \left[(1 - \tau)(1 - \psi) \frac{\beta s_1}{(1 + \beta s_1)} - \frac{g^A g^N \tau (1 - \psi)}{(1 + r)(1 + \beta s_1)} \right]$$

To have 1 + r on the left-hand-side, the expression can be rearranged to have:

$$\iota^{S} : r = \frac{\tau}{(1-\tau)\frac{\beta s_{1}}{g^{A}g^{N}} - \iota \frac{1+\beta s_{1}}{(1-\psi)[g^{A}g^{N}-(1-\delta)]}} - 1}$$
(D.15)

Assuming a PAYGO pension system with fixed replacement rate, the tax rate is given by (D.10) on a BGP: $\tau = \bar{d}/(\bar{d} - g^N/s_1)$.

E Solow residual in the euro area

The Solow residual is identified by the TFP component in the Cobb-Douglas production function employed in the OLG model:

$$Z_t = \frac{Y_t}{K_t^{\psi} L_t^{1-\psi}} \tag{E.1}$$

using data on output (Y_t) , capital (K_t) , labor (L_t) and imposing a value for the capital bias ψ . Recall that for the sake of the OLG model and economic interpretation only its growth rate matters: $g_t^Z = Z_t/Z_{t-1}$.

The main series are computed using the "Area Wide Model" (AWM) dataset (https:// eabcn.org/page/area-wide-model) which provides main macroeconomic series at the quarterly frequency for the euro area as a whole in the period 1970-2017.⁷¹ To have a value for 2018 the same procedure used on AMW data is applied on data from the "World Development Indicators" (WDI, wdi.worldbank.org) summing the series for the 12 euro area countries forming EA12 which are available for the period 1970-2018.⁷² The resulting Solow residual growth rates obtained employing the two dataset are essentially identical except for the 4 years before 1975 casting doubt on data reliability before that. To obviate this, for the period 1955-1975 the Solow residual growth rate comes from the the growth rate of "TFP at constant national prices (2011=1)" (rtfpna) provided by the "Penn World Table" (PWT, https: //www.rug.nl/ggdc/productivity/pwt/). For the period 1975-2018 the series on TFP growth provided by PWT and the one obtained applying (E.1) with AMW (and WDI) data are very similar in level and co-move very strongly with a Pearson's correlation coefficient of about 0.996. The series obtained from the AWM (and WDI) dataset for the period 1975-2018 is preferred due to its closer theoretical correspondence with the OLG model.

The data used to measure the annual Solow residual in the euro area for the period 1970-2017 are the following:

- *Output*. Annualized series YER ("Gross Domestic Product (GDP) at market prices, Million Euro, Chain linked volume, Calendar and seasonally adjusted data, Reference year 1995") in the AMW dataset.
- Capital. The capital stock in the euro area is built on the basis of the annualized series IER ("Gross Fixed Capital Formation, Millions of euros, Chain linked volume, Calendar and seasonally adjusted data, Reference year 1995") in the AMW dataset. Following standard practice (cf. e.g. Caselli (2005)) the perpetual inventory method is applied. The initial capital stock (the base year is 1970, the first year data are available) is computed using the formula: $K_{1970} = I_{1970}/(g_I + \delta)$ where I_{1970} corresponds to the gross capital formation in 1970, g_I is the average growth rate (about 1.657% in the period 1971-2017), while δ_K is set to 11% (the same value set in the OLG model calibration with non-zero TFP growth). The capital stock is obtained via the law-of-motion: $K_{t+1} = (1 - \delta)K_t + I_t$ for all years t from 1970 to 2017.

⁷¹For a description of this database, see Fagan et al. (2001).

⁷²EA12 is composed by the following countries: Austria (AT), Belgium (BE), Finland (FI), France (FR), Germany (DE), Greece (EL), Ireland (IE), Italy (IT), Luxembourg (LU), Netherlands (NL), Portugal (PT), Spain (ES).

• Labor. To be consistent with the OLG model data on labor are the same ones employed in the OLG model's simulations. That is, $L_t = h_j N_{t,j}$ measures the *effective* labor input where the profile h_j (Figure A.3) takes into account productivity and participation rates by age with exogenous retirement at age 65, and $N_{t,j}$ is the respective number of people for each year t for all ages j between 15 and 65 provided by UN (2017).

Setting the parameter ψ to 0.22 (to be consistent with the OLG model calibration) the series for output, capital and labor are used to measure the Solow residual Z_t in (E).

Figure E.1 plots the annual growth rate of the resulting Solow residual in the euro area used in the OLG model's simulations (recall that before 1975 the series comes from the estimates of TFP growth provided in the PWT). For each year in the period 1950-1954 the growth rate is simply set to the value in 1955 (the first year data are available). In the projected horizon it is assumed that the series reverts to its unconditional mean implied by an autoregressive process of order 1, AR(1), estimated over the period 1999-2018 as detailed in the note under the figure.



Figure E.1: Solow residual: annual growth rate, euro area

Note. "Actual" denotes the growth rate of the TFP component in (E) for the period 1975–2018, while for the period 1950–1974 the series is from the TFP growth provided by the PWT, as explained in the main text. It is assumed that in the projected horizon (after 2018) the series follows an AR(1) process $z_t = \theta_1 + \theta_2 z_{t-1}$ where θ_1, θ_2 are estimated over the 1999-2018 period. The estimation implies that in the long-run the series reverts to its unconditional mean: $\theta_1/(1 - \theta_2) = 0.00964$. "HP filtered" denotes the series resulting from the application of the Hodrick–Prescott filter with a multiplier equal to 1000 (a similar high value is used by Gagnon et al. (2016) in the context of OLG simulations for the US). In this case the series is assumed to revert to its unconditional mean beginning with year 2007, where this mean results from the estimation of an AR(1) process over the entire sample period 1951–2018.

F Measures of real interest rate in the euro area

To compare the model's simulation results with the data a set of different measures of the real interest rate is used (see Figure 11) for each of the EA12 countries.

From the dataset built by Jordá et al. (2019) (JST), available at http://www.macrohistory. net/data, the following measures of the yield component of the total return are used:

- bond_rate: "long-term interest rate", 10-year government bonds
- bill_rate: "short-term interest rate", i.e. "yield on Treasury bills, i.e., short-term, fixedincome government securities" or money market rate or deposit rate when T-bill unavailable).
- capital measured as the unweighted average of bond_rate, bill_rate, eq_dp ("equity dividend yield") and housing_rent_rtn ("housing rental return").

Each nominal return series x_t above, for each year t, is measured in real terms dividing by the annual variation of consumer prices p_t (cpi, "consumer prices (index, 1990=100)") such that the respective real interest rate is: $(1 + x_t)/(1 + \pi_t) - 1$, with $\pi_t = p_t/p_{t-1} - 1$.

Figure F.1 shows the capital series for those EA12 countries where series from the JST dataset were available.

A measure of the marginal product of capital (MPK) net of depreciation is derived applying the representative firm's first order condition in the OLG model:

$$r = MPK - \delta = \psi/(K/Y) - \delta \tag{F.1}$$

using data from the Penn World Table (PWT) national accounts, available at https://www. rug.nl/ggdc/productivity/pwt/,⁷³ on the capital-output ratio (K/Y) and the time-varying depreciation rate of capital (δ).⁷⁴ The series used for Y, K and δ are respectively: rgdpna ("Real GDP at constant 2011 national prices (in mil. 2011US\$)"), rnna ("Capital stock at constant 2011 national prices (in mil. 2011US\$)") and delta ("Average depreciation rate of the capital stock"). Since this depreciation rate tends to be lower than what calibrated in the OLG model (the average across the 12 countries over time in the PWT is about 3.5%) the capital bias in production ψ is set to the value of 0.28, higher than what assumed in the OLG model (0.22) in order to be consistent with the overall level of the real interest as found in the econometric estimate of HLW which is the main comparison reference for the OLG model. It comes as a consequence that the level of the marginal product of capital net of depreciation must be treated with caution.

Figure F.2 shows the resulting series of the marginal product of capital net of depreciation for the EA12 countries obtained using the PWT data. There is a marked negative trend common to most countries. The figure compares these series with the series for the marginal product of capital obtained using the standard method to measure capital employed in section E from AWM data for the euro area.⁷⁵ There is a marked discrepancy in the early years of the sample which, as already noted in section E, casts doubt on the reliability of AWM data in that part of the sample.

⁷³See Feenstra et al. (2015) for a description

⁷⁴As explained in Feenstra et al. (2015), in contrast to studies such as Caselli (2005) or Caselli and Feyrer (2007) (representative for the broader literature) the depreciation rate in the PWT varies across countries and over time as countries differ in the asset composition of their capital stock and depreciation differs across assets.

⁷⁵In this case, the real interest rate is still identified by (F.1) but the parameter values are those set in the OLG model $(\psi = 0.22 \text{ and } \delta = 0.11 \text{ for all periods}).$



Figure F.1: Real return on capital in the euro area

Note. Data source: JST. The return on capital is measured as the unweighted average of the yield component of the four assets returns available in the JST dataset: bond, bill, equity, housing.



Figure F.2: Marginal product of capital net of depreciation in the euro area *Note*. Data source: PWT. Luxembourg excluded from EA12 as evident outlier.

G Real interest rate: further comparison of the model results with data





Note. The long-run growth imposed in the final stationary equilibrium is denoted by $g_{LR} = (g^Z)^{1/(1-\psi)}$. See Appendix E for the Solow residual used in the simulations: actual and Hodrick-Prescott (HP) filtered. Econometric estimate for the euro area (EA) corresponding to the "natural rate of interest" in Holston et al. (2017) (HLW). The econometric estimate for the world economy is provided in Del Negro et al. (2019) (DGGT) on yield data provided by Jordá et al. (2019) (JST). Data from the same source are used for the darker shaded areas comparing the 90-10 percentile range of EA12 countries on real yields of different maturities and types (T-bill, T-bond, equity, housing). The brighter (yellow) shaded area shows the 90-10 percentile range of EA12 countries on a measure of the marginal product of capital (MPK) computed using data from the Penn World Table (PWT) with time varying depreciation rates. Details in Appendix F. The thin black lines mark a band corresponding to $g_{LR} = 1.7\% \pm 0.6\%$ in the OLG simulations.

H Robustness: alternative to perfect annuity market

If one assumes that there are no annuities, one needs to make an assumption about the allocation of the assets left by deceased households ("accidental bequest"). Following e.g. Börsch-Supan et al. (2014), here it is assumed that accidental bequests resulting from premature death are taxed by the government at a confiscatory rate and used for otherwise neutral government consumption. Under this assumption the representative household is not ensured against the risk of prematurely dying with positive wealth.⁷⁶ In this case, the budget constraint (2.3) in the main text needs to be substituted by:

$$a_{t+j+1,j+1} = a_{t+j,j}(1+r_{t+j}) - c_{t+j,j} + y_{t+j,j}$$
 (H.1)

Among the clearing conditions one needs to add government consumption G_t :

$$G_t = \sum_{j=0}^{J} a_{t,j} N_{t-1,j-1} (1 - s_{t,j}) (1 + r_t)$$
(H.2)

$$C_t + G_t + I_t = Y_t \tag{H.3}$$

Importantly, when one solves for the optimal conditions, in the Euler equations the conditional survival probabilities do not cancel out anymore. Hence, in addition to the new budget constraint, in steady sate the following new dynamics of consumption holds:

$$c_{0} = \frac{\sum_{j=0}^{J} \left(\frac{1}{1+r}\right)^{j} y_{j}}{\sum_{j=0}^{J} \left(\frac{1}{1+r}\right)^{j} \left(\beta(1+r)\right)^{\frac{j}{\sigma}} \pi_{j}^{\frac{1}{\sigma}}}$$
(H.4)

$$c_{j+1} = [\beta(1+r)s_{j+1}]^{\frac{1}{\sigma}}c_j, \text{ for } j = 0, 1, ..., J-1$$
 (H.5)

The analytic expressions for the rest of the model are unaffected by the new assumption on accidental bequest.

Figure H.1 plots the life-cycle profiles of consumption and assets holdings in the final steady state comparing the results obtained with (baseline) and without perfect annuity market. Since the conditional survival probabilities now enter the Euler equation, individual consumption is not increasing constantly over the life-cycle but exhibits a hump-shape pattern. Consistently with what found in Hansen and Imrohoroglu (2008), consumption declines at later stages of life because individuals are not compensated via annuities for their increasing effective rate of discount (due to survival probabilities falling as an individual ages). This decline starts happening when an individual's effective discount rate is larger than the interest rate.

The new simulation with no annuities is done assuming that the economy starts from the same value of the real interest rate in the initial steady state (0.0619). This requires to re-calibrate the discount factor β which results to be 0.9733 (instead of 0.9598 in the baseline). The remaining parameter values are the same of the baseline calibration (see Table 4.1). The real interest rate in the final steady state with no annuities is lower than in the baseline (about 3.15% per annum versus 4.55% in the baseline).

⁷⁶As noted e.g. by Quadrini and Ríos-Rull (2015), an alternative with identical implications, except for the use of public revenues, would be to assume that any household is like a pharaoh and assets are buried with their owners. For sensitivity on different degrees of annuitization of household's wealth, see Hansen and Imrohoroglu (2008).



Figure H.1: Final steady state: profiles of consumption and asset holdings

Note. Baseline calibration (see Table 4.1). In the "no perfect annuity" case the individual discount factor β is recalibrated in order to have the same real interest rate value in the initial steady state of the baseline. It results $\beta = 0.9733$.

Figure H.2 shows the results of the transition dynamics in the model for the main macroeconomic variables with (baseline) and without perfect annuities. Compared to the baseline, the absence of annuities increases the the willingness to save of individuals in the face of aging. The saving rate stands at a higher level throughout the whole transition, associated with a higher capitaloutput ratio and a lower real interest rate. Output growth benefits slightly, tending to stay at a slighter higher level. Overall, the economy is more sensitive to aging. So, while the saving rate is always at a higher level it decreases more in the acutest phase of the transition: between 1990 and 2040 (roughly the peak to trough) it decreases by about 2.9 percentage points (in the baseline about 2.15 percentage points over the same period). Similarly, the real interest rate decrease by about 1.55 percentage points between 1990 and 2030 (compared with 1.3 percentage points in the baseline). If anything, this exercise offers evidence that the downward impact of aging on the real interest rate in the baseline simulation is a conservative estimate.

Figure H.3 strengthens the comparison with the baseline by showing the the dynamics of the main aggregates in terms of deviations from the initial steady state until year 2200. Clearly, the case of no perfect annuities entails a a bigger downward pressure of aging on the real interest rate.



Figure H.2: EA12 transition dynamics: main macro variables

Note. For calibrations, see note of Figure H.1. Perfect foresight solution for the "no perfect annuities" case found via a homotopy method (see *Dynare* reference manual).



Figure H.3: EA12 transition dynamics: main aggregates per capita *Note*. See note of Figure H.2.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1280 The non-linear effects of the Fed asset purchases, by Alessio Anzuini (June 2020).
- N. 1281 The effects of shop opening hours deregulation: evidence from Italy, by Lucia Rizzica, Giacomo Roma and Gabriele Rovigatti (June 2020).
- N. 1282 *How do house prices respond to mortgage supply?*, by Guglielmo Barone, Francesco David, Guido de Blasio and Sauro Mocetti (June 2020).
- N. 1283 *The macroeconomics of hedging income shares*, by Adriana Grasso, Juan Passadore and Facundo Piguillem (June 2020).
- N. 1284 Uncertainty matters: evidence from a high-frequency identification strategy, by Piergiorgio Alessandri, Andrea Gazzani and Alejandro Vicondoa (June 2020).
- N. 1285 *The economic drivers of volatility and uncertainty*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (July 2020).
- N. 1286 *Protectionism and the effective lower bound in the euro area*, by Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (July 2020).
- N. 1287 *Court efficiency and aggregate productivity: the credit channel*, by Guzmán González-Torres and Giacomo Rodano (July 2020).
- N. 1288 *The time-varying risk of Italian GDP*, by Fabio Busetti, Michele Caivano, Davide Delle Monache and Claudia Pacella (July 2020).
- N. 1289 Trade shocks and credit reallocation, by Stefano Federico, Fadi Hassan and Veronica Rappoport (July 2020).
- N. 1290 The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions, by Raffaele Gallo (July 2020).
- N. 1291 An economic assessment of the evolution of the corporate tax system in Italy, by Ernesto Zangari (September 2020).
- N. 1292 Asymmetric information in corporate lending: evidence from SME bond markets, by Alessandra Iannamorelli, Stefano Nobili, Antonio Scalia and Luana Zaccaria (September 2020).
- N. 1293 *Fiscal space and the size of fiscal multiplier*, by Luca Metelli and Kevin Pallara (September 2020).
- N. 1294 Prudential policies, credit supply and house prices: evidence from Italy, by Pierluigi Bologna, Wanda Cornacchia and Maddalena Galardo (September 2020).
- N. 1295 How loose, how tight? A measure of monetary and fiscal stance for the euro area, by Nicoletta Batini, Alessandro Cantelmo, Giovanni Melina and Stefania Villa (September 2020).
- N. 1296 *Price dividend ratio and long-run stock returns: a score driven state space model*, by Davide Delle Monache, Ivan Petrella and Fabrizio Venditti (September 2020).
- N. 1297 *Workforce aging, pension reforms, and firm outcomes,* by Francesca Carta, Francesco D'Amuri and Till von Wachter (September 2020).
- N. 1298 Anti-poverty measures in Italy: a microsimulation analysis, by Nicola Curci, Giuseppe Grasso, Pasquale Recchia and Marco Savegnago (September 2020).
- N. 1299 Forecasting US recessions: the role of economic uncertainty, by Valerio Ercolani and Filippo Natoli (September 2020).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.
2018

- ACCETTURO A., V. DI GIACINTO, G. MICUCCI and M. PAGNINI, Geography, productivity and trade: does selection explain why some locations are more productive than others?, Journal of Regional Science, v. 58, 5, pp. 949-979, WP 910 (April 2013).
- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057 (March 2016)**.
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, WP 909 (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing*. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).

SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform*, European Journal of Political Economy, v. 59, pp. 423-444, **WP 1135 (September 2017).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, *Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis,* Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, **WP 1086 (October 2016).**
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area,* IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, WP 1186 (July 2018).
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme*, Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, **WP 1185 (July 2018).**
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, v. 22, 2, pp. 171-185, WP 1132 (September 2017).
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, Forecasting with instabilities: an application to DSGE models with financial frictions, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears, Economic Modelling, v. 77, pp. 266-275, WP 1076 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, **WP 836 (November 2011).**
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side* and demand-side factors, Social Indicators Research, v. 143, 3, pp. 1185–1217, **WP 989 (November** 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme*, IMF Economic Review, v. 67, 2, pp. 315-348, WP 1036 (October 2015).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, **WP 1167 (March 2018)**.
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*,1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, **WP 1127 (July 2017).**
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MERCATANTI A., T. MAKINEN and A. SILVESTRINI, *The role of financial factors for european corporate investment*, Journal of International Money and Finance, v. 96, pp. 246-258, **WP 1148 (October 2017).**
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, v. 46, 4, WP 984 (November 2014).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, **WP 1047 (January 2016).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times*, R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014).**
- BRIPI F., D. LOSCHIAVO and D. REVELLI, *Services trade and credit frictions: evidence with matched bank firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy*, The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, **WP 1299 (October 2020).**
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, **WP 1232 (July 2019).**
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., Raising aspirations and higher education. Evidence from the UK's widening participation policy, Journal of Labor Economics, v. 38, 1, pp. 183-214, WP 1188 (September 2018).
- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, WP 1072 (July 2016).

FORTHCOMING

- ALBANESE G., G. DE BLASIO and A. LOCATELLI, *Does EU regional policy promote local TFP growth? Evidence* from the Italian Mezzogiorno, Papers in Regional Science, WP 1253 (December 2019).
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, *Discretion and supplier selection in public procurement*, Journal of Law, Economics, and Organization, WP 1178 (June 2018).

- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, WP 1201 (November 2018).
- BOTTERO M., F. MEZZANOTTI and S. LENZU, Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy, Journal of International Economics, WP 1032 (October 2015).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, WP 1208 (February 2019).
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, WP 1156 (December 2017).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economics, **WP 1217 (April 2019).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area*, Open Economies Review, **WP 1162 (January 2018).**
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds, Annals of Applied Statistics, WP 1213 (March 2019).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, WP 1105 (March 2017).
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth, WP 1095 (January 2017).
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, **WP 1223 (June 2019).**
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility*, Journal of Human Resources, WP 1182 (July 2018).
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, **WP 1193 (September 2018).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, WP 1200 (November 2018).
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, WP 1092 (January 2017).
- PERICOLI M., On risk factors of the stock-bond correlation, International Finance, WP 1198 (November 2018).
- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, WP 1189 (September 2018).
- RAINONE E., The network nature of otc interest rates, Journal of Financial Markets, WP 1022 (July 2015).